Machine Learning techniques for Supply Chain Management: A Systematic Literature Review

Aicha El Filali, El Habib Ben Lahmer, Sanaa El Filali

Hassan II University, Casablanca, Morocco

aicha.elfilali-etu@etu.univh2c.ma

Abstract. This paper presents a systematic review of the development of research on Machine Learning (ML) applications in supply chain management (SCM), particularly demand forecasting. The objective is to bridge the gap between current and potential ML methods that can improve the performance of the demand forecasting operation in three SCM industries: industry, agriculture, and services, by analyzing published work in the last decade in several numerical databases. Five aspects were covered: the research trend on ML applications in SCM; the ML algorithms most used by researchers for demand forecasting; the advantages of ML algorithms over traditional forecasting approaches; the ML techniques considered most adopted at the demand forecasting operation, as well as the SCM industry sector that benefited most from these applications. We have focused on providing an overview of the applications of ML in demand forecasting and asserting its powerful role in the evolution of SCM effectiveness, in order to provide a reference for future research and applications of SCM in different economic sectors.

Keywords: Machine learning, supply chain management, demand forecasting, search trend, economic sectors

1. Introduction

In the current economic context, especially with the globalization; the interpenetration of markets; and the trend of outsourcing/offshoring, the company of the 21st century should be able to confront a new environment defined by (Bennett & Lemoine, 2014; Packowski, 2013) as the acronym VUCA (Volatility- Uncertainty-Complexity- Ambiguity). The supply chain, which incorporates the management of a company's processes from planning to distribution, it works to assure the optimal management of such processes in order to maintain a matching offer to the demand, is considered the first to be affected in this environment (Sasser, 1976; Chandra & Fisher, 1994; Mahmoud 2021).

According to (Milgrom & Roberts, 1988) uncertainty of demand is considered one of the most important components of its environmental, which increases the risk of mismatching between offer and demand. (Chen et al., 2007) suggest that demand uncertainty spills thrueghout processes, reducing the efficiency of the supply chain. (Spalanzani et al., 2006) define demand uncertainty as the inability to specify the exact quantities to produce in order to cover the markets needs at the specified time (timing of) of that demand. There are several actions to be adapted to the uncertainty of demand, the most known one is the increase in the level of stocks, which means, the production process (make to stock) (Milgrom & Roberts, 1988). However, this solution is not the best, because it rounds up the stationary costs related to the stock and affects the financial health of the company. There is another mechanism to reduce the imbalance between the offer and the demand, it is the improvement of the forecast of the demand. As (Haberleitner et al., 2010) argue, demand forecasting is the starting point for all planning activities and execution processes across the supply chain. Therefore, obtaining forecasts is a complicated process which need a great care in each full field phase, including the data retrieval and analysis step, which is the key to producing a good forecast.

The excessive use of technologies, associated with the usage of mobile internet and Internet of Things (IoT), have increased the data of several characteristics called Big Data. This latter is a data set distinguished with the five Vs: Volume, Velocity, Variety, Veracity, and Value (White, 2012; Jeble et al., 2017; Affia et al., 2019; Anu et al., 2021). Data analysis refers to the use of mathematical and statistical tools offering an available data analysis to extract symptomatic information helping in decision making. However, if the data continues growing simultaneously, it will be managed difficultly (Anu Dahiya et al., 2021). Traditional statistical methods failed to analyse these new and potentially invaluable data sets in order to generate good forecasts. (Jeble et al., 2017) confirmed that the presence of big data in the supply chain has created remarkable complexity in the demand forecasting operation.

Artificial intelligence has been one of the most popular techniques for many years, its power has been demonstrated in many applications (Min, 2010; Oke, 2004; beom kim, 2019). The definition of artificial intelligence is differentiated according to the

context in which it is used (Riahi et al., 2021), AI can be generally defined as « any artificially created intelligence, i.e. a software system that simulates human thinking on a computer or other devices » as defined (Čerka et al., 2015). The ability of AI-based automation technologies to manage and analyze large amounts of big data, has highlighted his role as a decision-making tool (Dwivedi et al., 2021), and has increased the creativity of the supply chain (Wilson & Daugherty, 2018) becoming more and more artificial for the company (Tarafdar et al., 2017). Machine learning is a subdomain of artificial intelligence able to learn from the data in order to produce predictions. According to many researches, Machine Learning can be applied to several stages of supply chain management, including demand forecasting as well as predictive analysis tasks (H. Bousqaoui et al., 2017; Raguseo, 2018). ML algorithms can interpret the complex non-linear relationships between so many causal factors affecting demand, in the presence of big data (Ampazis, 2014; Tanaka, 2010), and increase the efficiency of supply chain processes (Minis & Ampazis, 2006; Risi et al., 2018).

The importance of research on machine learning (ML) and supply chain management (SCM) has led to an excessive number of publications in recent years. Several researches have recently focused on the integration of machine learning techniques in the supply chain, such as (Bousqaoui et al., 2017; Feki et al., 2016; Mori et al., 2012), while others have described their applications for demand forecasting, such as (Waller & Fawcett, 2013; Carbonneau et al., 2008; Weng et al., 2019). Although machine learning has potential score, the number of publications in the area of demand forecasting remains modest. (Manyika et al., 2011) conforms that ML was applied in one or more SC activities in only 15% of firms. This could be due to a misunderstanding of ML concepts in the corporate culture, ambiguity of how it could be applied, knowledge of guidelines for selecting appropriate ML techniques for supply chain researchers and practitioners, as well as the inability to have appropriate data (Ni et al., 2020).

In fact, there are a few papers focused on systemic research of AI applications in SCM (Min, 2010; Riahi et al., 2021). (Ni et al., 2020) are the only ones to have established a review of ML applications in SCM, but their study was limited only to applications in the industry domain. Therefore, we have not found any reviews dedicated entirely to ML applications in demand forecasting.

For this purpose, this study attempts to complement the research of (Ni et al., 2020), by conducting a more in-depth systematic review of the applications of the 10 most widely used ML algorithms in demand forecasting, in the three main economic sectors: industrial sector, agricultural sector, and service sector in order to fill the gap between ML and SCM, and provide a comprehensive sample for researchers and practitioners in the field of SCM.

To reach this objective, we are called to establish a systematic review of the last ten years of publications in several major international databases, in order to answer five key questions of this research:

RQ1: What are the general research trends on applications in demand forecasting?

RQ2: Which ML algorithms are often used in demand forecasting?

RQ3: What are the advantages of ML algorithms compared with traditional approaches?

RQ4: Which *ML* algorithms are considered to be the most effective in demand forecasting?

RQ5: In which sectors and activities are these ML algorithms frequently used?

The rest of this paper will be organized as follows: section 2 presents the literature review about this study. Section 3 describe the Systematic review process followed in this study. Section 4, presents the results of analysis. Section 5 concludes, by proposals for future researchers.

2. Literature Review

2.1. Demand forecasting impact on the supply chain

Research in the field of operation management, has confirmed the role of effective supply chain management on the variability and continuity of many organizations in today's competitive market (Burinskiene 2018). (Fatorachian & Kazemi, 2018) points out that in order to keep pace with customer demands in a highly competitive environment, the supply chain should increase resilience and efficiency across processes. (Ivanov & Rozhkov, 2020) advocates that changes in demand profiles affect the operational politics of the supply chain by inducing disruptive queues on inventory dynamics. In recent years, world-class organizations have succeeded in always occupying the first place, because they have been able to interpret the expectations of its customers and follow the rapid evolution of technology, such as Google, Netflix, Airbnb, Amazon and Uber (Iansiti & Lakhani, 2020). Nevertheless, there are other organizations who had to stop their activities because they were not able to follow the technological trend and satisfy the demand of consumers, for example: Blockbuster in the United States. Thus, it can be deduced that the more efficient and responsive the supply chain is, the more competitive advantages the organization can gain. A critical factor in the effectiveness of supply chain management is the accuracy of demand forecasting (Chong et al., 2017). Several researches have discussed the effects of demand forecasting on supply chain performance, an accurate demand forecast synchronizes flows, ensures proper supply chain management and improves customer satisfaction by avoiding stock-outs (Kumar et al., 2020). (Fleischmann et al., 2002) considered demand estimation as a decision support tool, allowing the organization of the company's supply chain on all the time horizons and at all levels. (Oliva & Watson, 2009) approved that the

improvement of the demand forecasting mechanism allows the increase of the stock turnover rate and the reduction of 50% of the available stocks, the organization of the transport flows (Rai et al., 2019), and the growth of distribution activity (Shareef et al., 2016), by increasing financial benefits and return on investment (Gunasekaran et al., 2004; Clarke, 2006). While other researches have studied the negative effects of poor forecasting on the supply chain, (Lee et al., 1997; Wan & Evers, 2011; Mason-Jones & Towill, 1999) have shown that the order amplification called bullwhip effect (BWE) is among the most responsive effects of poor demand estimation. The (BWE) increases the overall cost and decreases the efficiency of the supply chain processes (Chopra & Meindl, 2007). Therefore, demand planning becomes one of the biggest challenges for companies (Moon et al., 2003; Mccarthy et al., 2006). In addition, in an effort to save production costs, companies have shifted to sourcing products from distant markets around the world (Christopher & Holweg, 2017). This, in turn, has caused an extension of both production and product sourcing times. According to (Armstrong, 1986; Simchi-levi & Zhao, 2005), as the forecasting horizon gets longer, the accuracy of demand forecasts tends to decrease. The quest for reliable customer demand forecasting is one of the most dynamic problems in supply chain management (Chong al., 2017). Traditional forecasting methods, such as moving average, time series, exponential smoothing, and Box-Jenkins methods, have relied heavily on the accuracy and consistency of historical data, which is not always the case in most of the time. They do not easily capture the non-linearity associated with time series data, as seasonality, trends, etc. Although historical data is always elementary for predicting future demand, there are multiple variables called dependent variables that are related to demand in a direct and indirect way and that must be taken into account during forecasting calculations. Nevertheless, traditional statistical methods are unable to capture non-linear relationships. For this reason, Supply Chain Management can leverage machine learning to overcome the limitations of statistical approaches to produce more reliable and efficient forecasting models.

2.2. Machine learning methods for demand forecasting

Demand forecasting is a discipline of predictive analytics that focuses on understanding customer demand for goods and services. In the field of demand forecasting, popular traditional models, including time series methods: ARIMA (Autoregressive Integrated Moving Average,) Moving average, weighted moving average, Croston model, etc... have dominated for a long time and have taken their place in many real-world applications. They are considered effective when the data series is stationary and takes a linear model. However, most real-world problems have nonlinear characteristics, drawing attention to nonlinear techniques (Aras & Kocakoç, 2016). Traditional methods can only generate results based on past and current data, without considering external demand factors as inputs. Therefore, Machine Learning (ML) methods can be a promising solution to address the limitations of traditional approaches and produce more reliable forecasts.

Machine learning, invented by Samuel (1959), emerged as a technique for allowing computers to acquire knowledge from data without being explicitly programmed (Ratner, 2000). ML models are able to capture non-linear relationships in data, without the limitations of time series. There are many different ML algorithms, which can generally be divided into two main categories: supervised and unsupervised learning. They are used to accomplish four main tasks: classification, regression, association and clustering (Kone & Karwan, 2011). In the case of supervised learning, the input data is labelled, that is to say, the model already knows what it is looking for in the data. During the learning process, the accuracy of the algorithm must also be adjusted to find the link between the inputs and outputs in order to achieve a good performance. Unlike supervised learning, unsupervised context must operate from unlabelled data. The role of the algorithm is to pursue cohesion within the data and to classify it into distinct groups as possible (Goodfellow et al., 2016). Machine learning algorithms are packages found in many popular programming languages, they were invented by researchers in the field, based on mathematical models built from data samples (beom kim, 2019). For example, Vapnik is the inventor of Support Vector Machine (SVM) and Huang guangbin is the inventor of Extreme Learning Machine (ELM) etc... They can be combined with other mathematical models, but their operating principles remain the same. (Ni et al., 2020) are the only ones to have conducted a review systematic review of the articles published during the period 1998/01/01-2018/12/31, on six academic databases: Emerald Insight, IEEE Xplore, Scopus, Science Direct, Wiley and Springer, with Google Scholar. The results showed that demand forecasting is the largest beneficiary of ML applications among the 6 SCM activities, and there is 10 ML algorithms which are the frequently used in the field of supply chain: Extreme Learning Machine (ELM), Random Forest (RF), k-nearest neighbor (KNN), Decision Tree (DT), Logistic Regression (LR), K-means , Naive Bayes Classifier (NBC), Neural Networks, Ensemble Algorithms (ESM). Among these algorithms, there are Neural Network (NN) methods and Support Vector Machine (SVM) which gather the most number of works carried on demand forecasting. For this reason, this study attempts to complement the research of (Ni et al., 2020), by conducting a more in-depth systematic review of the applications of the 10 most widely used ML algorithms in demand forecasting, in the three main economic sectors: industrial sector, agricultural sector, and service sector.

3. Systematic Review Process

This research focuses on the literature review of past and current applications of machine learning techniques in the field of demand forecasting. In our literature research, we followed the Systematic Literature Review method (SLR), as a well-

adapted methodology to searching, classifying and exploring scientific articles from digital databases (Brereton et al., 2007; Kitchenham, 2004). A systematic review is the rigorous and reproducible synthesis of the results of all existing original studies addressing the same research question (Colombet, 2015), it is an in-depth research process that reduces the risk of duplication of articles, making it more efficient than traditional research (Tranfield et al., 2003), SLR generally involves three successive stages: planning, elaboration, and presentation of results. The planning part consists of determining the keywords of the topic being addressed and rephrasing them as a research question, and the digital sources on which we will address our research. The elaboration of the methodology concerns the study of the selected articles which contains the stage of the definitions of the criteria of inclusion and exclusion, and the stage of the Quality Assessment where we must determine Quality assessment questions and finally the stage of extraction of the necessary data for our study. Finally, we have the results part where we present all the results of this systemic study.



Fig. 1: Stages of review protocol

Applying these steps, the string (machine learning OR linear regression OR neural network Extreme Learning Machine OR Random Forest OR k-nearest Neighbor OR Decision Tree OR Logistic Regression OR K-means OR Naive Bayes Classifier OR Neural Networks OR Support Vector Machine OR Ensemble Algorithms) AND (demand forecasting) was used to retrieve articles from three of Science digital databases Scopus (www.scopus.com) and Web (www.webofscience.com), IEE Xplore (www.ieeexplore. ieee.org). These tree databases are considered among the largest sources of various types of documentations, namely scientific journals, books and conferences, and they allow

to dedicate a universal view of the progress of scientific and technological research. The search for articles was limited to the last 10 years, from 2010 to 2020. We limited our query to journals and conferences with a good academic reputation. The first list of articles retrieved from the search strings was 1522 articles, as shown in Figure 1: 600 from Scopus, 651 from Web of Science, and 271 from IEE Xplore as a first selection, and according to our determined inclusion and exclusion criteria we kept 315 articles, then, after downloading the articles, we submitted them to a quality assessment the number of articles becomes 176.

3.1. Justification of the selection process

We refined the selection of studies by reading the titles, abstracts, and keywords. Implementation of the inclusion and exclusion criteria on our first selection reduced the total number to 315 articles.

We defined the following inclusion and exclusion criteria:

3.1.1. Inclusion criteria

The following inclusion criteria was established: (i) journals and conference papers dealing with the intersection of demand forecasting and Machine Learning algorithms, containing the search terms in the title, abstract or keywords (ii) journals and conference papers written in English; (iii) journals and conference papers published since 2010.

3.1.2. Exclusion criteria

For the quality assessment of the study, the following exclusion criteria was defined for: (i) papers using the term machine learning outside of demand forecasting, (ii) papers without evaluation criteria for the forecasting methods used, (iii) papers containing the keywords in the abstract just to present the study context; (iiii) paper without electronic document.

3.1.3. Quality assessment

Quality assessment, is an operation that requires a comprehensive review of the articles selected in the previous steps, there is no universal definition of parameters of the quality of the studies to be followed, but it has been defined by the fact that the biases are the minimum possible and the internal and external validation is maximum (Centre for Reviews & Dissemination, 2001). Distinguishing the studies that answer our research questions (RQ) and supporting the process of inclusion/exclusion are the main reasons for which the quality assessment was carried out. Quality assessment is suggested as a tool to enhance individual work as it allows us to interpret the results obtained and establish comparative analyses, for future research direction in the field of software engineering (Brereton et al., 2007). Following the list proposed by Kitchenham our quality assessment (QA) questions are specified below:

No.	Question
QA1	Does the study define the context in which the research was established?
QA2	Are the objectives of the research adequately defined?
QA3	Does the study identify the research questions?
QA4	Is the research plan in line with the research objectives?
QA5	Is the literature review appropriate to the research question?
QA6	Have the supply chain sector and the type of demand to be predicted been well defined?
QA7	Is the source of the data sets indicated?
QA8	Can the data collected address the research question?
QA9	Are the prediction techniques used justified?
QA10	Does the study specify the best performing method?
QA11	Are there clear statements of results and conclusions?
QA12	Are the limitations of the research addressed?
QA13	Is the study useful for research or practice?

Table 1: Quality assessment questions

Each quality assessment question will be scored according to three possible options: (1) the problem is completely addressed, (0.5) partially addressed, (0) no longer addressed. Then, a quality score will be calculated by summing the scores of the responses on each quality question. The quality assessment of the studies relevant to this research was performed, and disagreements on the results were discussed among the researchers, in order to retain only those with a quality score. The articles are subsequently classified, according to the calculation of scores into four categories: Excellent ($12 \le \text{score} \le 13$, Good ($10,5 \le \text{score} \le 11.5$), Medium ($9,5 \le \text{score} \le 10$), Low ($6.5 \le \text{score} \le 9$), Very low ($4,5 \le \text{score} \le 7$).

Quality level	# Of studies	Percent
Excellent $(12 \le \text{score} \le 13)$	64	33%
Good $(10,5 \le \text{score} \le 11.5)$	86	45%
Medium $(9,5 \le \text{score} \le 10)$	26	13%
Low $(6.5 \le \text{score} \le 9)$	14	7%
Very low $(4,5 \le \text{score} \le 7)$	3	2%
Total	193	100%

Table 2: Relevant study quality measurement

Following the quality assessment measure, 17 articles with a Low and very Low score were excluded from our final list. We finally end up with 176 qualified articles for the data extraction stage.

3.2. Data extraction

The extraction strategy for the selection of studies, we required that the quality score be higher than 9. Indeed, as shown in Table 2, we believe that the selected studies are of high quality since about 94% (184 out of 193) of the selected studies are of

excellent or good or Medium quality level. In order to collect the data we need to answer the research questions of this study, we first examined the content of each article and then designed a form in the form of a card (Table 3) to organize and facilitate the pilot extraction of data. The elements of the table are distinguished (source of publication; year of publication of the article; authors; target industry of the research; type of demand to be forecasted, motivation for the use of machine learning; research method; sample of the experiment; ML methods used; evaluation factors of the chosen forecasting model; impacts and benefits of the application of machine learning on demand forecasting in the targeted domain).

All these elements are classified and grouped, in such a way, to answer each research question. The extracted data were rechecked and discussed among the researchers before filling in the form for each item that will be used later for a data synthesis.

Table 3. Data extraction form

Source of publication
Year of publication of the article
Authors
Sector of activity targeted by the research
Type of demand to be predicted
Sample of the experiment
ML methods used
Evaluation factors of the selected forecasting model
Impacts of the chosen ML model application on the target domain
RQ1: What are the general research trends on applications in demand forecasting?
The year of publication of the articles allows us to group them by year which can
show us the growth or the decrease of research on ML applications in demand
forecasting.
PO2: Which ML algorithms are often used in demand forecasting?
ML techniques most appropriete for producing effective demand forecasts
will techniques most appropriate for producing effective demand forecasts
RO3: What are the advantages of ML algorithms compared to traditional approaches?
Comparison between ML vs. non-ML models
Sample of the experiment
strengths of ML algorithms
suchgate of the algorithms
RO4: Which ML algorithms are considered to be the most effective in demand
forecasting?
Type of demand to forecast
Comparison between ML models
1
RQ5: In which sectors and activities are these ML algorithms frequently used?
Target industry of the research
Impacts of the application of the selected ML model on the targeted domain

4. Results and Discussion

4.1. General research trends on ML applications in demand forecasting (RQ1)

Fig. 2, shows the trend of papers published over the last ten years on machine Learning and demand forecasting. We observe that there is a remarkable fluctuation in the last three years. The 76 papers published in 2018 and half of 2021 alone accounted for 48% of the total publications over the past ten years. This shows that ML holds more popularity in the SC domain, especially in demand forecasting.



Fig. 2: Number of papers published on machine learning between 2010 and 2021

Fig. 3, presents the evolution of research during the last ten years on ML and SCM on the Internet, according to Google Trends, which is a tool that allows to follow the trends related to a defined topic. We can see that our results are compatible with those of Google Trends, where the frequency of searches on ML has had a significant jump in the last three years.



Fig. 3: Google research on the growth of interest in machine learning and supply since 2010

4.2. Identification of frequently used algorithms in demand forecasting (RQ2)

From the selected papers, we identified eleven ML methods that were applied to estimate demand forecasts. They are presented below:

- Artificial Neural Networks (ANN)
- Support Vector Machin (SVM)
- Random Forest (RF)
- Extreme Learning Machine (ELM)
- EXtreme Gradient Boosting (XGBoost)
- k-Nearest Neighbor (K-NN)
- Decision Tree (DT)
- Linear Regression (LR)
- Naïve Bayes Classification (NBC)
- Ensemble learning (ESM)
- Genetic algorithm (AG)

The analysis of the identified papers shows that neural networks and support vector machine are the most used algorithms, with 53% and 21% of the total papers reviewed, which shows the popularity of these algorithms in this field. The other algorithms are between 1% and 5%, however, these algorithms are able to produce good forecasts, but it may simply be due to the lack of popularity for demand forecasting in each of these three areas. The details of research attention to each ML technique over the past decade are presented in Table 4.

Several researchers have tested the demand forecasting performance of these less popular algorithms against the most frequently used ones. For example, (Huang et al., 2019) have compared four learning methods, support vector regression (SVR), extreme machine learning (ELM), EXtreme Gradient Boosting (XGBoost), and linear regression (LR) for energy demand forecasting of residential buildings. The authors have observed that (ELM) provides the best performing results by reducing the error rate on the forecasts. Similarly, (Jin et al., 2020) proposed a hybrid model based on the Kernel extreme learning machine (KELM), variational mode decomposition (VMD), autoregressive moving average (ARMA) model for short-term forecasting of the number of air passengers in Beijing, Guangzhou and Pudong airports. The performance of this model was compared with the hybrid model composed with Support Vector Machine (SVM). The test showed the strength of the Extreme Learning Machine (ELM) based algorithm. As a result, a primordial call is made to researchers in the field of supply chain to give more importance to the less popular machine learning algorithms in the different economic sectors.

4.3. ML algorithms vs traditional approaches (RQ3)

Since ML is a data-driven technique, the construction and the model chosen are based on the nature and size of the data in the ML project. Thus the evaluation of the ML model depends on the dataset. In addition to the dataset the metrics used to measure the accuracy of the ML model must be taken into consideration. There are different metrics for measuring the accuracy of ML methods, so it is important to choose the appropriate metrics to evaluate the performance of the predictions. In this study, the selected studies showed that correlation coefficient, Root Mean Square Error (RMSE), Mean Average Percentage Error (MAPE), and absolute error (MAD) are the most popular evaluation metrics to measure the accuracy and performance of developed models.

Machine		Number
Learning	Scientific article reference	of
Algorithms		articles
Artificial Neural Network (ANN)	(Kumar et al., 2020); (V1jat & Sivakumar, 2018); (Runge et al., 2020); (Uyar et al., 2016); (Liu et al., 2021); (Hošovský et al., 2021); (Y-C. Hu, 2020); (Yi Yang et al., 2016); (Abba & Elkiran, 2017); (Wu et al., 2021); (Ofori-Ntow Jnr et al., 2021); (Fu & Chien, 2019); (Hribar et al., 2019); (Sharifzadeh et al., 2019); (Szul et al., 2020); (Prado et al., 2020); (Haq et al., 2019); (Kantasa-ard et al., 2020); (Chen et al., 2021); (Bedi & Toshniwal, 2018); (Domingos, Ojeme, & Daramola, 2021); (Jintao Ke et al., 2017); (Chong, Han, & Park 2017); (Ribeiro et al. 2020); (He 2017); (H. Wang et al. 2018); (Kilimci et al., 2019); (C. Zhang et al., 2020); (Ayub et al., 2020); (G. Zhang et al., 2020); (Asijad et al., 2020); (G. Zhang et al., 2020); (Chang et al., 2020); (Abbasimehr et al., 2020); (J. Liu et al., 2020); (X. Li et al., 2020); (Chang et al., 2011); (Chaturvedi et al., 2013); (Kaytez et al., 2015); (Lau et al, 2012); (Romano & Kapelan, 2014); (Shahrabi et al., 2012); (Romano & Kapelan, 2014); (Shahrabi et al., 2012); (Venkatesh et al., 2014); (C. Xia et al., 2015); (C. Chang et al., 2012); (Chen & Wang, 2012); (Du et al., 2015); (Catal et al., 2012); (Chen & Wang, 2012); (Du shuay-Vila et al., 2012); (Chen & Wang, 2012); (Du et al., 2015); (Catal et al., 2015); (C. Cia et al., 2016); (Catal et al., 2017); (Simani et al., 2016); (C. Liu et al., 2014); (C. Xia et al., 2015); (Catal et al., 2016); (C. Liu et al., 2014); (C. Xia et al., 2015); (Chen et al., 2017); (Simani et al., 2016); (Catal et al., 2016); (Marinescu et al., 2014); (Hossen et al., 2016); (Catal et al., 2017); (Varanasi & Tripathi, 2016); (Ramos-Carrasco et al., 2017); (Ai et al., 2019); (Delorme-Costil & Bezian, 2017); (Varanasi & Tripathi, 2016); (Ramos-Carrasco et al., 2017); (Ai et al., 2019); (Chandi et al., 2021); (UNUTMAZ et al., 2020); (Kanavos et al., 2021); (Carianesi, et al., 2020); (Selvi & Mishra, 2020); (Al-Ghamdi et al., 2021); (Laroussi, et al., 2020); (Chandi et al., 2021); (Chandi et al., 2021); (Chandi et al., 2021); (Chan	108

Table 4: Summary of publications on machine learning algorithms

	Do, 2017) ; (Guo et al., 2021) ; (Hamzacebi et al., 2019) ; (Y C. Hu et al., 2019) ; (Z. Huang et al., 2019) ; (Chauhan & Hanmandlu, 2010) ; (Ma & Luo, 2021) ; (Bin & Tianli, 2020) ; (Das, 2017) ; (Kankal & Uzlu, 2017); (Shilaja & Arunprasath, 2020); (Ao, 2011); (Germi et al., 2014) ; (Feizabadi, 2020); (Souza, Wanke, & Correa, 2020)	
Support Vector Machin (SVM)	(Al-Musaylh et al., 2018); (Priyadarshi et al., 2019); (Aktepe et al., 2021); (Allawi et al., 2019); (Beyca et al., 2019); (Solyali, 2020); (Jawad et al., 2020); (Abolghasemi et al., 2020); (Jawad et al., 2020); (Abolghasemi et al., 2020); (LY. Chen, 2014) ; (WC. Hong et al., 2011) ; (Z. Hu et al., 2014) ; (Lu, 2014) ; (Yeh et al., 2014) ; (X. Yu et al., 2013) ; (WC. Hong et al., 2013) ; (Sarhani & El Afia, 2015); (Villegas et al., 2018); (Idowu et al., 2016); (Eseye & Lehtonen, 2020); (Abera & Khedkar, 2020) ; (Bolandnazar et al., 2020); (Y. Fu et al., 2015); (Gonçalves et al., 2021); (Nuaimi, 2014); (Vink et al., 2020); (Y. Wang, 2011); (Qiao et al., 2019) ; (M. S. Li et al., 2015) ; (Xue et al., 2018) ; (Tabrizchi et al., 2021) ; (Vahdani et al., 2016) ; (Fan et al., 2020); (Fan et al., 2021)	33
Random Forest (RF)	(Punia et al., 2020); (Johannesen et al., 2019); (Qiu et al., 2017) ; (Herrera et al., 2010) ; (Lahouar & Ben Hadj Slama, 2015) ; (J. Kim et al., 2018) ; (Kang et al., 2017) ; (Yin et al., 2020);	8
Extreme Learning Machine (ELM)	(Jin et al., 2020); (J. Kumar & Singh, 2020) ; (Choi et al., 2014) ; (M. Xia et al., 2012) ; (Y. Huang et al., 2019) ;	5
EXtreme Gradient Boosting (XGBoost)	(Moroff et al., 2021); (Ji et al., 2019) ; (Almaghrebi et al., 2020) ; (Sidhu et al., 2020)	4
k-Nearest Neighbor (K-NN)	(Panagopoulos et al., 2020) ; (Al-Qahtani & Crone, 2013) ;	2
Decision Tree (DT)	(Van Nguyen et al., 2020) ; (Bozkir & Sezer, 2011); (Laik et al., 2014);	3
Linear Regression (LR)	(Zhu et al., 2018); (Merkuryeva et al., 2019); (Amin et al., 2019); (Ciulla & D'Amico, 2019) ; (Demiriz, 2014); (Yandong Yang, 2015); (Francis & Kusiak, 2017); (H. Lee et al., 2014) ; (T. Hong et al., 2010) ; (Akpinar & Yumusak, 2017)	10
Naïve Bayes Classificatio n (NBC)	(Chu, 2014)	1
Ensemble learning (ESM)	(Vanichrujee et al., 2018)	1
Genetic algorithm (AG)	(Ghanbari et al., 2013) ; (Azadeh et al., 2014)	2

The review of the selected papers showed that, based on the evaluation metrics, the researchers demonstrated that machine learning methods can produce better forecasts compared to traditional SCM demand forecasting methods. We can cite, (Hošovský et al., 2021) compared weekly gas demand forecasts produced by traditional ARMA and SARMA models and a genetic algorithm-optimized regression wavelet neural network (REGWANN) ML method, for three buildings with different profiles. The researchers used the daily gas consumption in the three buildings over a five-year period. The researchers showed using the mean absolute error (MAE) measures the superiority of the ML model, then that it allowed the reduction of the error values by 22.6%, 17.7% and 57% for the three buildings. Similarly, (Moroff et al., 2021) who applied two statistical time series forecasting methods namely seasonal auto regressive integrated moving average (SARIMAX), Triple exponential smoothing (ETS) and three ML methods Random Forest (RF), Extreme Gradient Boosting (XGBoost), Long short term memories (LSTM), for forecasting the demand of five products, in the area of industry 4.0, using six years' sales history of each product. The performance of the forecasts is measured using RMSE values, where the ML models produced the most accurate demand forecasts than traditional methods, with even lower error rates for the (LSTM) method. Also, (Lau et al., 2013) applied artificial neural network (ANN) and exponential smoothing and multiple regressions, for gasoline demand forecasting. The researchers used historical monthly gasoline demand in millions of gallons from Ontario from 1960 to 1975. Using the MSE, the researchers showed that the ANN model was able to produce the most optimal forecast.

The advantages of ML algorithms are diverse, such as the ability to learn from a large amount of multidimensional data, the ease of manipulation and the support for decision making in a shorter time compared to traditional approaches. This conclusion has been supported by several studies reported in the literature, where researchers have compared the predictions produced by the ML method and traditional approaches. From our list of selected articles, we can cite: (Gonçalves et al., 2021) automotive electronics assembly product demand, (Moroff, Kurt, & Kamphues, 2021; Ji et al., 2019) E-commerce product demand, (Kantasa-ard et al., 2020) agricultural SC internet network demand, (A. Kumar, Shankar, & Aljohani, 2020) electronic product demand, (Abbasimehr, Shabani, & Yousefi, 2020) furniture demand etc.

Other researchers such as (Fradinata et al. 2019; Jaipuria & Mahapatra 2014) have studied the ability of ML techniques to reduce the bullwhip effect in SCM. The authors confirmed that more accurate predictions produced lower amplification of the bullwhip effect across members of a supply chain.

4.4. Performance comparison between ML algorithms in demand forecasting (RQ4)

The Fig. 3, shows that 61% of the researches done over the last ten years, applied neural networks, followed by the support vector machine algorithm at 19%. This result is consistent with several systemic studies under the same research axis (Aamer, Eka Yani, & Alan Priyatna, 2020; EL Filali et al., 2021; Ni et al., 2020; Bousqaoui et al., 2017).



Fig. 3: Distribution of article publications by machine learning method

This can be interpreted by the fact that most researchers use NNs to compare the performance of predictions, such as (Solyali, 2020) who have applied four ML algorithms: artificial neural network (ANN), adaptive neuro-fuzzy inference system (ANFIS), support vector machine (SVM), and multiple linear regression (MLR), on real historical data of electricity consumption in 2016 and 2107, for the electricity demand forecasting in the long and short run. Temperature, humidity, solar irradiation, population, gross national income (GNI) per capita, and electricity price per kilowatthour were chosen as the input parameters for the ML algorithms. The results of the long-term and short-term studies show that SVM and ANN are more reliable than other ML methods, however ANN model is the most superior in terms of accuracy. (Sharifzadeh, Sikinioti-Lock, & Shah, 2019), have compared Artificial Neural Networks (ANN), Support Vector Regression (SVR) and Gaussian Process Regression (GPR), for solar, wind and electric energy demand prediction. The authors applied a sensitivity analysis for each algorithm to adjust its parameters and obtain the most optimal results. All models were able to produce good predictions of solar and wind energy demand, but only the neural network method was able to predict electricity demand. Similarly for (Hribar et al., 2019) have compared the residential natural gas demand forecast of an urban area produced by : Linear regression (LR), Recurrent neural networks (ANN) and the extreme learning machine. The results show that accuracy is much higher when the model learns the forecasted weather data and Recurrent Neural Networks (ANN) were found to be the best performers for demand forecasting in this energy sector. Other researchers have compared the

forecasts produced by the artificial neural network (ANN) algorithm versus the support vector machine (SVM), as in the paper by (Abera & Khedkar, 2020), the authors have applied both algorithms to forecast the electrical demand of individual household electronics, and they showed that the SVM provided more accurate and much better forecasting results than previous work in the same area. Similarly, for (Beyca et al., 2019) who applied multiple linear regression (MLR), artificial neural network (ANN) and support vector regression (SVR) for natural gas demand forecasting, in the largest natural gas consuming megacity Istanbul province in Turkey. The results showed that SVR provided the most reliable and accurate results compared to ANN technique. Other researchers have chosen to combine NN with other statistical or ML models in order to further improve forecasting performance, as in the case of (Barzegar et al., 2020; Sajjad et al., 2020; J. Ke et al., 2017; Ayub et al., 2020; Mohammadi et al., 2014). The unbalanced distribution of ML algorithms, is common for items from all databases used in this study, as stated (Ni et al., 2020; Bousqaoui et al., 2017). Other algorithms have just appeared, such as ELM which is frequently used for decision support in the financial domain (M.-J. Kim, 2012), nevertheless, it is little applied in the SCM domain, this can be translated by the ambiguity carried on these algorithms regarding the over-interpretation of data which can cause catastrophic results in this domain. In general, in answering question 4, the analysis of our study confirms that NNs are the most appropriate methods to produce efficient demand forecasts in the SCM domain.

4.5. ML models by economic sector and ML application area (RQ5)

Table 5, contains a total of 176 articles identified through our systematic search. Each article is classified according to the field of application of the supply chain, and affected to one of three sectors of economic activity, namely, the industry sector, the service sector and the agriculture sector. In order to give a global view of the economic sectors and under sectors most benefited by the applications of new technologies.

Machine Learning Algorithm	Reference	Total articles	% algorithm	% Sub- sector	% Sector
	Industry sector				66%
<u>R</u>	esidential gas demand	5		4%	
Artificial Neural Network (ANN)	(Hribar et al., 2019); (Hošovský et al., 2021)	2	40%		
Support Vector Machine (SVM)	(Beyca et al., 2019) ; (Qiao et al., 2019)	2	40%		
Linear Regression (LR)	(Akpinar & Yumusak, 2017)	1	20%		
	Crude oil demand	1		1%	
Artificial Neural Network (ANN)	(Shao et al., 2014)	1	100%		

Table. 5. Distribution of articles by sector of activity and Machine Learning algorithm used

	Water demand	12		10%	
Artificial Neural Network (ANN)	(Vijai et al., 2018) ; (Nasser et al., 2020); (Romano & Kapelan, 2014); (Delorme-Costil & Bezian, 2017) ; (Ren et al., 2010); (Al-Ghamdi et al., 2021) ; (Y. Xu et al., 2019) ; (Chen et al., 2019) ;	8	67%		
Support Vector Machine (SVM)	(Allawi et al., 2019)	1	8%		
Random Forest (RF)	(Herrera et al., 2010);	1	8%		
XGBoost	(Sidhu, Kumar, & Rana, 2020)	1	8%		
	Electricity demand	48		41%	
Artificial Neural Network (ANN)	(Runge et al., 2020) ; (Yang et al., 2016) ; (Ofori-Ntow Jnr et al., 2021) ; (Wang et al., 2018) ; (Ayub et al., 2020); (Chang et al., 2011); (Chaturvedi et al., 2015); (Jaramillo-Morán et al., 2013); (Kaytez et al., 2015); (Li et al., 2013); (Li et al., 2014); (Lou & Dong, 2015); (Niu et al., 2012); (Unsihuay-Vila et al., 2010); (Xia et al., 2010); (Yadav & Srinivasan, 2011); (Kotillova et al., 2012); (Sarhani & El Afia, 2015); (Haque et al., 2012); (Sajjad et al., 2020); (Daniel et al., 2016) ; (Bedi & Toshniwal, 2018) ; (C. Liu et al., 2017); (Sahay et al., 2016); (Marinescu et al., 2014); (Hossen et al., 2018); (Selvi & Mishra, 2020) ; (JF. Chen et al., 2017) ; (Hamzacebi et al., 2019) ; (Chauhan & Hanmandlu, 2010) ; (Germi et al., 2014);	31	65%		
Support Vector Machine (SVM)	(Al-Musaylh et al., 2018); (Solyali, 2020);(Hu et al., 2014); (Hong et al., 2013); (Fu et al., 2015); (Jawad et al., 2020); (Vink et al., 2020); (M. S. Li et al., 2015); (Fan et al., 2020); (Villegas et al., 2018); (Zhu et al. 2018); (T. Hong et al.	10	21%		
(LR)	(2010) ; (Lee et al., 2014)	3	6%		
k-Nearest Neighbor (K-NN)	(Al-Qahtani & Crone, 2013)	1	2%		
Random Forest (RF)	(Lahouar & Ben Hadj Slama, 2015); (Yin et al., 2020);	2	4%		
Algorithme génétique	(Azadeh et al., 2014)	1	2%		
	Energy demand			25%	
Artificial Neural Network (ANN)	(Sharifzadeh et al., 2019); (Szul et al., 2020); (Prado et al., 2020); (Ribeiro et al., 2020); (Zhang et al., 2020); (G. Zhang et al., 2020); (Yeh et al., 2014);	17	59%		

	(Chen & Wang, 2012); (Cheng & Wei,				
	2010); (Haq et al., 2020); (He, 2017);				
	(Ai, et al., 2019) ; (Varanasi & Tripathi,				
	2016); (UNUTMAZ et al., 2021);				
	(Ahmad et al., 2020) ; (Kankal & Uzlu,				
	2017); (Shilaja & Arunprasath, 2020);				
	(Das, 2017)				
Support Vector	(Abera & Khedkar, 2020); (Eseye &				
Machine (SVM)	Lehtonen, 2020) ; (Tabrizchi et al.,,	4	14%		
	2021) ; (Idowu et al., 2016)				
XGBoost	(Almaghrebi et al., 2020)	1	3%		
Random Forest	(X. Qiu et al., 2017) ; (Johannesen et	2	7%		
(RF)	al., 2019)	_	. , •		
Extreme Learning	(Huang et al. 2019)	1	3%		
Machine (ELM)	(Truing et un, 2017)	-	570		
k-Nearest Neighbor	(Panagopoulos et al., 2020)	1	3%		
(K-NN)	(Fundgopoulos et un, 2020)	1	570		
Algorithme	(Ghanhari et al. 2013)	1	3%		
génetique	(Ghanbari et al., 2013)	1	570		
Linear Regression	(Ciulla & D'Amico, 2019) ; (Amin et	2	7%		
(LR)	al., 2019)	2	7 70		
	dry freight demand	1		1%	
Artificial Neural	(Uvar et al 2016)	1	100%		
Network (ANN)	(Oyai et al., 2010)	1	100%		
Mag	nesium material demand	1		1%	
Artificial Neural	(11 2020)	1	1000/		
Network (ANN)	(Hu, 2020)	1	100%		
Oxygen deman	d of the wastewater treatment plant	1		1%	
Artificial Neural		1	1000/		
Network (ANN)	(Abba & Elkiran, 2017)	1	100%		
Se	emi-conductor demand	1		1%	
Artificial Neural			100-1		
Network (ANN)	(Fu & Chien, 2019)	1	100%		
	Goods demand	1		2%	
Artificial Neural		-	100-1		
Network (ANN)	(Weng et al., 2019) ; (Feizabadi, 2020)	2	100%		
	Furniture demand	1		1%	
Artificial Neural				- / •	
Network (ANN)	(Abbasimehr et al., 2020)	1	100%		
Airci	aft spare engines demand	1		1%	
Artificial Neural				170	
Network (ANN)	(Liu et al., 2020)	1	100%		
Constru	iction machinery spare parts	3		3%	
Linear Regression				- / 0	
(LR)	(Francis & Kusiak, 2017)	1	33%		
Random Forest					
(RF)	(J. Kim, Lee, & Choi, 2018)	1	33%		
Support Vector					
Machine (SVM)	(Aktepe et al., 2021)	1	33%		
Pharm	aceutical products demand	1		1%	
1 1141 111	and a producto definition		1	÷ / V	

Linear Regression	(Merkurveva et al., 2019)	1	100%		
(LR)	Beauty Demand	1		1%	
Fuzzy inference		<u> </u>	1000/	1 /0	
system	(Souza et al., 2020)	1	100%		
Automobile ele	ectronic assembly products demand	2		2%	
Linear Regression (LR)	(Gonçalves et al., 2021)	1	50%		
Artificial Neural Network (ANN)	(Chandriah & Naraganahalli, 2021)	1	50%		
Clo	othing industry demand	5		4%	
Artificial Neural Network (ANN)	(Kilimci et al., 2019); (Liu et al., 2021); (Du et al., 2015)	3	60%		
Extreme Learning Machine (ELM)	(Choi et al., 2014); (Xia et al., 2012)	2	40%		
<u>Co</u>	smetic product demand	1		1%	
Support Vector Machine (SVM)	(Vahdani, Razavi, & Mousavi, 2016)	1	100%		
	Workload demand	1		1%	
Extreme Learning Machine (ELM)	(Kumar & Singh, 2020)	1	100%		
	Agriculture sector	4			2%
Ener	gy demand in agriculture	2		50%	
Artificial Neural Network (ANN)	(Saini et al., 2020)	1	50%		
Support Vector Machine (SVM)	(Bolandnazar et al., 2020)	1	50%		
Agri	cultural product demand	1		25%	
Support Vector Machine (SVM)	(Priyadarshi et al.,2019)	1	100%		
Physical Internet n	etworks of the agricultural SC demand	1		25%	
Artificial Neural Network (ANN)	(Kantasa-ard et al., 2020)	1	100%		
	Services Sector				32%
	Transport demand			18%	
Artificial Neural Network (ANN)	(Wu et al., 2021); (Gao & Lee, 2019); (J. Xu et al., 2018); (A. Y. Chen et al., 2016) ; (Toqué et al., 2017) ; (Vanichrujee et al., 2018) ; (Bejarano- Luque et al., 2021); (Z. Huang et al., 2019)	8	80%		
Random Forest (RF)	(Kang et al., 2017)	1	10%		
Linear Regression (LR)	1	10%			
New	spaper/magazine demand	1		2%	
Support Vector Machine (SVM)	(Yu et al., 2013)	1	100%		
	Passenger demand	<u>7</u>		12%	

			1		
Artificial Neural Network (ANN)	(Ke et al., 2017); (Li et al., 2020) ; (Laaroussi et al., 2020) ; (Kanavos et al. 2021)	4	57%		
Extreme Learning Machine (ELM)	(Jin et al., 2020)	1	14%		
Support Vector Machine (SVM)	(Wang, 2011)	1	14%		
Decision Tree (DT)	(Laik et al., 2014)	1	14%		
onli	ne movie rental demand	1		2%	
Artificial Neural Network (ANN)	(Ampazis, 2014)	1	100%		
E-co	mmerce product demand	4		7%	
Artificial Neural Network (ANN)	(Lau et al.,2013) ; (Ao, 2011)	2	50%		
XGBoost	(Moroff et al., 2021) ; (Ji et al., 2019)	2	50%		
F	ood products demand	1		2%	
Support Vector Machine (SVM)	(Abolghasemi et al., 2020)	1	100%		
	Tourism demand	7		18%	
Artificial Neural Network (ANN)	(Chen et al., 2012); (Shahrabi et al., 2013); (Teixeira & Fernandes, 2012); (Ramos-Carrasco et al. 2019); (Livieris et al., 2019) ; (Shi, 2020) ; (YC. Hu et al., 2019)	7	70%		
Support Vector Machin (SVM)	(Hong et al., 2011); (Fan et al., 2021)	2	20%		
Classificateur Naive Bayes (NBC)	(Chu, 2014)	1	10%		
I	Future export demand			4%	
Artificial Neural Network (ANN)	(Dave et al., 2021); (Bin & Tianli, 2020)	2	100%		
	Retail demand	5		9%	
Random Forest (RF)	(Punia et al., 2020)	1	20%		
Neural Network (NN)	(Slimani et al., 2015); (Wanchoo, 2019), (Feizabadi, 2020)	2	40%		
Support Vector Machine (SVM)	(Xue et al., 2018)	1	20%		
Linear Regression (LR)	(Demiriz, 2014)	1	20%		
Ele	ctronic Product demand	5		9%	
Decision Tree (DT)	(Van Nguyen et al., 2020)	1	20%		
Support Vector Machine (SVM)	(Chen, 2014); (Lu, 2014)	2	40%		
Artificial Neural Network (ANN)	(Lu et al., 2012);(Kumar et al., 2020)	2	40%		
	Food demand	4		7%	
Decision Tree (DT)	(Bozkir and Sezer, 2011)	1	25%		
Artificial Neural Network (ANN)	(L. Huang et al., 2021) ; (Guo et al., 2021) ; (Ma & Luo, 2021)	3	75%		

Filali et al, Journal of System and Management Sciences, Vol. 12 (2022) No. 2, pp. 79-136

Ho	tel reservation demand			2%	
Artificial Neural Network (ANN)	(J. Wang & Duggasani, 2020)	1	100%		
Hea	Ithcare Service Demand			2%	
Support Vector Regression (SVR)	(Nuaimi, 2014)	1	100%		
En	nergency items demand			2%	
Artificial Neural Network (ANN)	(Mohammadi et al.,2014)	1	100%		
<u>c</u>	<u>ash demand at ATMs</u>	3		5%	
Artificial Neural Network (ANN)	(Venkatesh et al., 2014); (Catal et al., 2015); (Domingos et al., 2021) ;	3	100%		
5	stock market demand	1		2%	
Artificial Neural Network (ANN)	(Chong et al., 2017)	1	100%		
	Total	176			100%

The results of our analysis as shown in Table 5, shows that (66%) of the research works on ML applications for demand forecasting are concentrated in the industrial sector, where electricity and energy demand forecasting hold more importance in the industrial sector with (41%) and (25%), respectively. Followed by water forecasting (10%) and clothing industry demand (4%), residential gas demand (4%), construction machinery spare parts (3%). However, a small number of papers studied the demand forecast of pharmaceutical industry (1%), automotive industry (2%), Beauty demand (1%), Cosmetic product demand (1%) and Workload demand (1%). According to (Ni, Xiao, & Lim 2020), the lack of appropriate and sufficient data needed for the success of these applications, ambiguity on the guidelines to be followed by CS practitioners allowing the selection of appropriate ML methods in order to improve demand forecasting, are among the main factors for the low applications of ML in companies in the industrial domain. ANNs are the most used algorithms in the industrial sector, over the last ten years, with a percentage of 60%, followed by SVM algorithms (17%). This result is consistent with several systemic studies under the same research axis (Aamer, Eka Yani, & Alan Priyatna, 2020; Ni, Xiao, & Lim, 2020).

The service sector is in second place with a percentage of 32%, due to the increase in mobile technologies, excessive use of the Internet and online services. The service sector occupies the second place with a percentage of 42%, due to the increase of mobile technologies, excessive use of the Internet and online services. We can find more ML applications for demand forecasting in tourism 18%, transportation 25%, e-commerce products 8%, electronic products 13%, cash demand in ATMs 8%, and retail 5%. As in the industrial sector, ANN algorithms are among the most used with a percentage of 65%, there are also regression algorithms 44% such as support vector regression 16%, decision tree (DT) (5%)and Random forest 4% which have also been used for supervised and unsupervised demand forecasting in the service sector, This is due to the development of technologies that allow the collection of data retrieved

through various tools, websites and connected objects and online applications used by customers, allowing the aggregation of sales history according to the categories of purchases made.

The results of our analysis showed that there is only 2% of research work in the field of ML in the Agriculture sector. This is an alarming figure in this important sector due to the low use of new technologies in the agricultural industry, this calls for researchers and supply chain practitioners to take advantage of the benefits of ML techniques for demand forecasting to improve the agricultural sector.

4.6. Research challenges and future research directions

Our study contributes to an in-depth investigation of the applications of ML techniques in SCM and more specifically in demand forecasting in three leading economic sectors: industrial, agricultural and service sectors. 176 articles were retrieved from three digital databases, Scopus, Web of Science and IEE Xplore, and then divided according to the year of publication, the ML method applied and the application domain.

Although we were able to answer the research questions as expected, the study has some limitations. First, only three databases were taken as sources. It is likely that there are other relevant articles to consider. For this reason, other academic databases should be included. Second, the 10 ML algorithms considered in this study are based on the systemic review established by (Ni al., 2020) in the field of SCM, there might be other low frequency ML algorithms in SCM, but they might be really useful for SCM. To this end, the list of algorithms should be updated regularly. Third, based on our systemic study, one of the trends of ML applications in forecasting is the dominance of neural network methods in 61% of the applications, this could be due to the ability of neural networks to produce more accurate forecasts compared to other algorithms such as linear regression (LR) and decision tree (DT). This result does not imply the ultimate superiority of neural networks, it all depends on the nature of the data to be processed and the context of the application (Goodfellow, Bengio, & Courville, 2016). 66% of the analyzed works were found in the industrial sector, 41% between these applications were concentrated in the field of electricity and very low publications in the automotive industry 2%, and the pharmaceutical industry 1%. The service sector occupies 32% of the publications and very low applications in the agricultural sector 2%. This number is alarming for an important sector such as agriculture. It is therefore necessary to call on researchers to deepen research in the SCM of the agricultural sector, as this is necessary for the economic development of countries.

From our analysis, most of the works have established comparison applications between ML methods with each other's or with statistical methods, in order to show the performance of ML techniques without proposing a strategy for integrating these techniques in the real world or studying the effects of these applications on the effectiveness of SCM. (Kilimci et al., 2019) are the only ones who have proposed an actual strategy for including a deep learning model in SCM.

Considering the disruption of the global economy and supply chain due to the COVID-19 pandemic, and drought, research in SCM should also be conducted in the service sector, and in the pharmaceutical and water industry to establish effective and transparent collaborative forecasting and replenishment strategies along the supply chain.

5. Conclusion

This study adopted a systematic review of recent trends in ML applications in SCM to answer the five research questions presented at the beginning of this study. The research articles in this study published in the last ten years were searched in three major academic databases: Scopus, Web of Science, and IEE Xplore. As a result, 176 articles were selected and closely analyzed. The result of the analysis showed that ML applications in SCM have experienced a remarkable increase in the last three years, and these methods have demonstrated their capabilities to provide more accurate forecasts at lower costs compared to traditional demand forecasting approaches. The distribution of papers by type of ML algorithm applied, showed a remarkable imbalance among the 10 frequently used ML algorithms in the SCM field. Neural network methods are the most dominant in 61% of the articles studied, followed by SVMs 33%. Similarly, the industrial sector receives the most attention from machine learning researchers where electricity and energy demand forecasting occupy a more important place with (41%) and (25%) respectively in this sector, while 4% of research was devoted to the automotive industry and 1% to the pharmaceutical industry. The service sector has the second largest number of applications with 32%, while the agricultural sector has the lowest number of searches (2%). We believe that machine learning can play an important role in improving efficiency throughout the supply chain of organizations. For this reason, this seminal study, can be a reference of what has been done in the three economic sectors for researchers and policy makers to move from statistical methods to ML methods to improve SCM efficiency for the three economic sectors.

Reference	Q A1	Q A2	Q A3	Q A 4	Q A 5	Q A6	Q A 7	Q A 8	Q A 9	Q A 10	QA 11	QA 12	QA 13	Total score
(Kaytez <i>et al.</i> , 2015)	1	1	0	1	1	1	1	1	0.5	1	0.5	0	1	10
(Lau <i>et al.</i> , 2013)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Li <i>et</i> <i>al.</i> ,2013)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Li <i>et al.,</i> 2014)	1	1	0	1	1	1	1	1	1	1	1	1	1	12

Table 6: Quality score of studies

(Lou & Dong, 2015)	1	1	0	1	1	1	1	0. 5	0.5	0.5	1	0	1	9.5
(Lu <i>et al.</i> , 2012)	1	1	1	1	1	1	1	1	1	1	1	1	1	13
(Lu, 2014)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Mohammadi et al., 2014)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Niu <i>et al.</i> , 2012)	1	0.5	0	1	1	1	1	1	1	1	1	0	0.5	10
(Romano & Kapelan, 2014)	1	1	1	1	1	1	1	1	1	1	1	1	1	13
(Shahrabi <i>et al.</i> , 2013)	1	1	1	1	1	1	1	1	1	1	1	1	1	13
(Teixeira & Fernandes, 2012)	1	1	1	1	0. 5	1	1	0	0.5	1	1	0	0.5	9.5
(Unsihuay- Vila <i>et al.,</i> 2010)	1	1	0	0. 5	1	1	1	1	1	1	1	0	1	10,5
(Venkatesh <i>et al.</i> , 2014)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Xia <i>et al.</i> , 2010)	1	1	0	1	1	0.5	1	1	1	1	1	0	1	10,5
(Xia <i>et al.</i> , 2012)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Yandong Yang, 2015)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Yeh <i>et al.</i> , 2014)	1	1	0	1	0	1	1	1	1	1	1	0	1	11
(Yu <i>et al.,</i> 2013)	1	1	0	0. 5	1	1	1	0. 5	0.5	1	1	0	1	9.5
(Bakker <i>et al.</i> , 2013)	1	1	0	0. 5	0. 5	1	1	1	1	0.5	0.5	0	1	9
(Catal <i>et al.</i> , 2015)	1	1	0	0. 5	1	1	1	1	1	1	1	0.5	1	11
(Chen & Wang, 2012)	1	1	0	1	1	1	1	0. 5	1	1	1	0.5	1	11
(Du <i>et al.,</i> 2015)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Ghanbari <i>et al.</i> , 2013)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Hassan <i>et al.</i> , 2015)	0.5	0.5	0	1	1	1	1	0. 5	1	1	0.5	0	0.5	8.5
(Hong <i>et al.</i> , 2013)	1	1	0	0. 5	1	1	1	1	1	1	1	1	1	11.5
(Joseph & Larrain, 2012)	1	0.5	0	0. 5	0. 5	1	1	1	0.5	1	1	0	1	9
(Yadav & Srinivasan, 2011)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Cheng & Wei, 2010)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Hassan, <i>et al.</i> , 2013)	1	0.5	0	0. 5	1	1	1	1	1	1	1	0.5	1	10.5

(Kotillova, <i>et</i> <i>al.</i> , 2012)	1	0.5	0	0. 5	1	0.5	1	1	0.5	1	1	0.5	1	9.5
(Sarhani and Fl Afia 2015)	1	1	0	1	1	1	1	0.	1	1	1	0	1	10,5
(Abba & Elkiran 2017)	1	1	0	0. 5	0. 5	1	1	0. 5	1	1	1	0	1	9.5
(Abbasimehr, <i>at al.</i> 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Abera &														
Khedkar, 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Abolghasemi <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Ahmad & Chen, 2019)	1	1	0	0. 5	1	1	0. 5	0. 5	1	0.5	0.5	0	1	8.5
(Aktepe <i>et al.</i> , 2021)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11.5
(Allawi <i>et al.</i> , 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Almaghrebi et al., 2020)	1	1	1	1	1	1	1	1	1	1	1	1	1	13
(Al-Musaylh et al., 2018)	1	1	0	1	1	0	1	1	1	1	1	1	1	12
(Al-Qahtani & Crone, 2013)	1	1	0	0. 5	1	0	1	1	1	1	1	1	1	11.5
(Amin <i>et al.</i> , 2019)	1	1	0	0. 5	1	0	1	1	1	1	0.5	1	1	10
(Ampazis, 2014)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Ayub <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11.5
(Azadeh <i>et</i> <i>al.</i> , 2014)	1	1	0	1	1	1	1	1	1	0.5	1	0	1	10.5
(Bedi & Toshniwal, 2018)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11.5
(Beyca <i>et al.</i> , 2019)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Bolandnazar, et al., 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Bozkir and Sezer, 2011)	1	1	0	1	0. 5	0.5	0. 5	1	0.5	1	1	1	1	10
(Boriratrit <i>et</i> <i>al.</i> , 2018)	1	1	0	1	1	0.5	0	1	0	0	1	0	1	7.5
(Cankurt, 2016)	1	1	0	1	0	0.5	1	0. 5	0	0	1	0	1	7
(Chang <i>et al.</i> , 2011)	1	1	0	1	1	1	1	1	1	0.5	1	0.5	1	11
(Chaturvedi, et al., 2015)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Chen <i>et al.</i> , 2012)	1	1	0	1	1	1	1	1	1	1	1	0	1	11

(Chen et al.	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Chen, 2014)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11.5
(Choi <i>et al.</i> , 2014)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Chong <i>et al.</i> , 2017)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Chu, 2014)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Ciulla & D'Amico, 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Dave <i>et al.</i> , 2021)	1	1	0	0. 5	1	0.5	1	1	1	1	1	1	1	11
(Demiriz, 2014)	1	1	0	1	0. 5	1	1	1	0.5	1	1	0	1	10
(Domingos, <i>et</i> <i>al.</i> , 2021)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Eseye & Lehtonen, 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Francis & Kusiak, 2017)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Fu & Chien, 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Fu <i>et al.</i> , 2015)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Gao & Lee, 2019)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Gonçalves <i>et</i> <i>al.</i> , 2021)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Guo <i>et al.</i> , 2017)	0.5	1	0	1	0. 5	0.5	0. 5	1	1	1	1	0	0.5	8.5
(Haq <i>et al.</i> , 2020)	1	1	0	1	0. 5	1	0. 5	0. 5	1	1	1	1	1	10.5
(Haque et al. 2012)	1	1	0	1	0. 5	1	0. 5	1	1	1	1	0	1	10
(He, 2017)	1	1	0	1	0. 5	1	1	1	1	0.5	1	0	1	10.5
(Herrera <i>et al.</i> , 2010)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Hong <i>et al.</i> , 2011)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Hošovský <i>et</i> <i>al.</i> , 2021)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Hribar <i>et al.</i> , 2019)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Hu, 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Hu <i>et al.,</i> 2014)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Huang <i>et al.</i> , 2019)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Idowu <i>et al.,</i> 2016)	1	1	0	1	1	1	1	1	1	1	1	0	1	11

(Jaramillo- Morán, <i>et al.</i> , 2013)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Jawad <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Ji <i>et al.</i> , 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Jin <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Johannesen, <i>et al.</i> , 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Kantasa-ard et al., 2020)	1	1	0	1	1	1	1	1	1	0.5	0.5	0	1	10
(Ke <i>et al.</i> , 2017)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11.5
(Kilimci <i>et al.</i> , 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Kumar <i>et al.,</i> 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Kumar & Singh, 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Lahouar & Ben Hadj Slama, 2015)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Laik, <i>et al.</i> , 2014)	1	1	0	0. 5	0. 5	1	1	1	1	0.5	1	0	1	9.5
(Lee <i>et al.</i> , 2014)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11.5
(Li <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11,5
(Liu <i>et al.,</i> 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Liu <i>et al.</i> , 2021)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Merkuryeva et al., 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Moroff <i>et al.</i> , 2021)	1	1	1	1	1	1	1	1	1	1	1	1	1	13
(Nasser <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Nuaimi, 2014)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11.5
(Ofori-Ntow Jnr, Ziggah, & Relvas 2021	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Ozerdem <i>et</i> <i>al.</i> , 2017)	1	0.5	0	0. 5	0. 5	0.5	1	1	1	1	1	0	1	9
(Panagopoulo s <i>et al.</i> , 2020)	1	1	0	1	1	0.5	1	1	1	1	1	0	1	10.5
(Prado <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Priyadarshi et al., 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12

(Punia <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Qiu <i>et al.</i> , 2017)	1	1	0	0. 5	1	1	1	0. 5	1	1	1	1	1	11
(Ribeiro <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Runge <i>et al.,</i> 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Solyali, 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Sharifzadeh, et al., 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Sajjad <i>et al.,</i> 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Szul <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Shao <i>et al.</i> , 2014)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Uyar <i>et al.,</i> 2016)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Vijai & Bagavathi Sivakumar, 2018)	1	1	0	1	1	1	1	1	0.5	0.5	1	0	1	10
(Villegas <i>et al.</i> , 2018)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Vhatkar & Dias, 2016)	0.5	0.5	0	1	1	1	1	1	1	0.5	0.5	0	0	8
(Van Nguyen, 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Vink <i>et al.,</i> 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Wang <i>et al.</i> , 2018)	1	1	1	1	1	1	1	1	1	1	1	1	1	13
(Weng <i>et al.</i> , 2019)	1	1	1	1	1	1	1	1	1	1	1	1	1	13
(Wu <i>et al.</i> , 2021)	1	1	1	1	1	1	1	1	1	1	1	0	1	12
(Wang, <i>et al.</i> , 2015)	1	0.5	0	0	0. 5	1	1	1	0.5	0.5	1	0	1	8
(Wang, 2011)	1	1	0	0. 5	0. 5	1	1	1	1	1	1	0	1	10
(Yang <i>et al.</i> , 2016)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Yin <i>et al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Zhu <i>et al.</i> , 2018)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(G. Zhang <i>et</i> <i>al.</i> , 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Zhang <i>et al.,</i> 2020)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Daniel, Kasun, & Milos 2016)	1	1	0	1	1	1	1	1	1	1	0.5	0	1	10.5

(I.V. (1									1					
(J. Xu et al. 2018)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(T. Hong et al. 2010)	1	0	0	1	1	1	1	1	1	1	1	0.5	1	10.5
(A. Y. Chen et al. 2016)	1	1	1	1	1	1	1	1	1	1	1	1	1	13
(Qiao et al. 2019)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Hamid & Rahman 2010)	1	0.5	0	1	1	0.5	1	1	0.5	0.5	0	0.5	0.5	8
(Ray, Mishra, & Lenka 2016)	1	0	0	1	1	0.5	0	1	1	1	0.5	0	0.5	7.5
(C. Liu et al. 2017)	1	0	1	1	1	1	1	1	1	1	1	1	1	12
(Slimani, El Farissi, & Achchab 2015);	1	1	0	1	0. 5	1	1	1	1	1	1	0	1	10.5
(Sahay, Sahu, & Singh 2016)	1	1	0	1	1	1	1	1	1	0	0	0.5	1	9.5
(Marinescu et al. 2014)	1	1	0	1	1	1	1	1	1	1	1	1	1	11
(Hossen et al. 2018)	1	1	0	1	1	0.5	1	1	1	1	0.5	0	1	10
(Bala 2010)	1	1	0	1	1	0.5	0. 5	0. 5	1	1	0.5	0	0.5	8.5
(Toqué et al. 2017)	1	1	0	1	1	1	1	1	1	1	0.5	0	1	10.5
(M. S. Li et al. 2015)	1	1	0	1	1	1	1	1	1	1	0.5	0	1	10.5
(Ai, Chakravorty, & Rong 2019)	1	1	0	0. 5	1	1	1	1	1	1	1	0	1	10,5
(Delorme- Costil & Bezian 2017)	1	1	0	0. 5	0. 5	1	1	1	1	1	1	1	1	11
(Singh, Singh, & Paliwal 2016)	1	0.5	0	0. 5	0. 5	1	0. 5	1	1	1	0.5	1	0.5	9
(Vanichrujee et al. 2018)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11.5
(Varanasi & Tripathi 2016)	1	0.5	0	1	1	1	1	1	1	1	0.5	0.5	1	10.5
(Ramos- Carrasco et al. 2019)	1	1	0	1	1	1	1	1	1	1	0.5	0.5	1	11
(Ren et al. 2010)	1	1	0	1	1	1	1	1	1	0.5	0.5	0	1	10
(Bejarano- Luque et al. 2021)	1	1	0	1	1	1	1	1	1	1	1	1	1	12

(H. Yu et al. 2021)	1	1	0	1	1	1	0. 5	1	0.5	0.5	0	0	1	8.5
(UNUTMAZ et al. 2021)	1	1	0	1	1	1	1	1	1	0.5	1	0	1	10.5
(Weisbach et al. 2020)	1	1	0	0. 5	0. 5	1	1	1	0	O. 5	0.5	0	1	8
(Xue et al. 2018)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Wanchoo 2019)	1	1	0	1	1	1	1	1	0.5	0.5	1	0	1	10
(Selvi & Mishra 2020)	1	1	0	1	1	1	1	1	0.5	0.5	1	1	1	11
(Al-Ghamdi, Kamel, & Khayyat 2021)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Laaroussi, Guerouate, & sbihi 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Kanavos et al. 2021)	1	1	0	1	1	1	1	1	1	1	1	0.5	1	11.5
(Feizabadi 2020)	1	1	1	1	1	1	1	1	1	1	1	1	1	13
(Livieris et al. 2019)	1	0.5	0	1	0. 5	1	1	1	0.5	1	1	0	1	10
(J. Wang & Duggasani 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Y. Xu et al. 2019)	1	0.5	0	1	1	1	1	1	1	1	1	0	1	11
(Ahmad et al. 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Shi 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Tabrizchi, Javidi, & Amirzadeh 2021)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Sidhu, Kumar, & Rana 2020)	1	1	0	1	0. 5	1	0. 5	1	0.5	1	1	0	1	9.5
(L. Huang et al. 2021)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Chandriah & Naraganahalli 2021)	1	1	0	1	1	1	1	1	1	1	0.5	0	1	10.5
(JF. Chen, Lo, & Do 2017)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Guo et al. 2021)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Hamzacebi, Avni, & Cakmak 2019)	1	1	0	1	1	1	1	1	0.5	0.5	0.5	0	1	9.5

(YC. Hu, Jiang, & Lee 2019)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Chauhan & Hanmandlu 2010)	1	1	0	0. 5	1	1	1	1	0.5	0.5	1	0	1	9.5
(Ma & Luo 2021)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Bin & Tianli 2020)	1	1	0	1	0. 5	1	1	1	1	1	1	0	1	10.5
(Das 2017)	1	1	0	1	0. 5	1	1	1	1	1	1	0	1	10.5
(Kankal & Uzlu 2017)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Shilaja & Arunprasath 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Olmedo 2016)	1	0.5	0	0. 5	0. 5	1	1	1	0	1	1	0	1	8.5
(Ao 2011)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Germi et al. 2014)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Vahdani, Razavi, & Mousavi 2016)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Souza, Wanke, & Correa 2020)	1	1	0	1	1	1	1	1	1	0.5	0.5	0	1	10
(Fan et al. 2020)	1	1	0	1	1	1	1	1	1	1	1	0	1	11
(Fan, Jin, & Hong 2021)	1	1	0	1	1	1	1	1	1	1	1	1	1	12
(Saini et al. 2020)	1	0.5	0	1	1	1	1	1	1	0.5	0.5	0	1	9.5
(Akpinar & Yumusak 2017)	1	1	0	1	0. 5	0.5	1	1	1	1	1	0	1	10

Acknowledgments

We would like to thank our professors at the Faculty of Science Ben M'sick, Mr. EL Habib Ben Lahmar and Mrs. Sanaa EL FILALI for their insightful guidance throughout this work. We also thank, Zadouani Hicham the director of the company Copimad for his warm support during the construction of this work, Mr. Saidou Ahmed, for his valuable advice, and Mr. Berji Hassan for his suggestions on the English presentation of this paper. This research was supported by the "Center for Innovation and Technology Transfer" of Hassan 2 University, Casablanca, Morocco. We would like to thank the president of Hassan II University, Ms. Mohammed Talbi, and the Director of the Center, Ms. Sanaa El Filali, as well as the entire team, for their

efforts to orient the research towards innovation and digital transformations for the prefectures of the greater Casablanca area.

References

Aamer, A., Yani, L. P. E. & Priyatna, I. A. (2020). Data analytics in the supply chain management: review of machine learning applications in demand forecasting. *Operations and Supply Chain Management: An International Journal* 14 (1), 1–13.

Abba, S. I. & Elkiran, G. (2017). Effluent prediction of chemical oxygen demand from the astewater treatment plant using artificial neural network application. *Procedia Computer Science*, 9th International Conference on Theory and Application of Soft Computing, Computing with Words and Perception, ICSCCW 2017, 22-23 August 2017, Budapest, Hungary, 120 (January), 156–63. https://doi.org/10.1016/j.procs.2017.11.223.

Abbasimehr, H., Shabani, M. & Yousefi, M. (2020). An optimized model using lstm network for demand forecasting. *Computers & Industrial Engineering* 143 (May): 106435. https://doi.org/10.1016/j.cie.2020.106435.

Abera, F. Z. & Khedkar, V. (2020). Machine learning approach electric appliance consumption and peak demand forecasting of residential customers using smart meter data. *Wireless Personal Communications* 111 (1), 65–82. https://doi.org/10.1007/s11277-019-06845-6.

Abolghasemi, M., Beh, E., Tarr, G. & R. Gerlach. (2020). Demand forecasting in supply chain: the impact of demand volatility in the presence of promotion. *Computers and Industrial Engineering* 142. https://doi.org/10.1016/j.cie.2020.106380.

Affia, I., Yani, L. P. E. & Aamer, A. (2019). Factors affecting iot adoption in food supply chain management. In.

Ahmad, J., Tahir, A. Larijani, H., Ahmed, F., Shah, S. A., Hall, A. J. & Buchanan, W. J. (2020). Energy demand forecasting of buildings using random neural networks. *Journal of Intelligent & Fuzzy Systems*, 38(4), 4753–65. https://doi.org/10.3233/JIFS-191458.

Ai, S., Chakravorty, A. & Rong, C. (2019). Evolutionary ensemble LSTM based household peak demand prediction. In 2019 International Conference on Artificial Intelligence in Information and Communication (ICAIIC), 1–6. https://doi.org/10.1109/ICAIIC.2019.8668971.

Akpinar, M. & Yumusak, N. (2017). Naive forecasting of household natural gas consumption with sliding window approach. *Turkish Journal of Electrical*

Engineering and Computer Sciences, 25(1), 30–45. https://doi.org/10.3906/elk-1404-378.

Aktepe, A., Yanik, E. & Ersöz, S. (2021). Demand forecasting application with regression and artificial intelligence methods in a construction machinery company. *Journal of Intelligent Manufacturing*. https://doi.org/10.1007/s10845-021-01737-8.

Al-Ghamdi, Al-Batool, Kamel, S. & Khayyat, M. (2021). Evaluation of artificial neural networks performance using various normalization methods for water demand forecasting. In *2021 National Computing Colleges Conference (NCCC)*, 1–6. https://doi.org/10.1109/NCCC49330.2021.9428856.

Allawi, M. F., Jaafar, O., Hamzah, F. M. & El-Shafie, A. (2019). Novel reservoir system simulation procedure for gap minimization between water supply and demand. *Journal of Cleaner Production*, 206, 928–43. https://doi.org/10.1016/j.jclepro.2018.09.237.

Almaghrebi, A., Aljuheshi, F., Rafaie, M., James, K. & Alahmad, M. (2020). Datadriven charging demand prediction at public charging stations using supervised machine learning regression methods. *Energies* 13(6). https://doi.org/10.3390/en13164231.

Al-Musaylh, M. S., Deo, R. C., Adamowski, J. F. & Li, Y. (2018). Short-term electricity demand forecasting with mars, svr and arima models using aggregated demand data in queensland, Australia. *Advanced Engineering Informatics*, 35, 1–16. https://doi.org/10.1016/j.aei.2017.11.002.

Al-Qahtani, F. H. & Crone, S. F. (2013). Multivariate k-nearest neighbour regression for time series data - A novel algorithm for forecasting UK electricity demand. In . https://doi.org/10.1109/IJCNN.2013.6706742.

Amin, P., Cherkasova, L., Aitken, R. & Kache, V. (2019). Analysis and demand forecasting of residential energy consumption at multiple time scales. In , 494–99.

Ampazis, N. (2014). Forecasting demand in supply chain using machine learning algorithms. *Journal of Artificial Life Research*, 5 (January). https://doi.org/10.4018/IJALR.2015010104.

Anu, D., Gautam, N. & Gautam, P. K. (2021). Data mining methods and techniques for online customer review analysis: A literature review. *Journal of System and Management Sciences*, December. https://doi.org/10.33168/JSMS.2021.0401.

Ao, S. I. (2011). A hybrid neural network cybernetic system for quantifying crossmarket dynamics and business forecasting. *Soft Computing*, 15(6), 1041–53. https://doi.org/10.1007/s00500-010-0580-4. Aras, S. & Kocakoç, I. D. (2016). A new model selection strategy in time series forecasting with artificial neural networks: IHTS." *Neurocomputing*, 174 (January), 974–87. https://doi.org/10.1016/j.neucom.2015.10.036.

Armstrong, J. (1986). Research on forecasting: A quarter-century review, 1960-1984." *Interfaces*, February, 89–109. https://doi.org/10.1287/inte.16.1.89.

Ayub, N., Irfan, M., Awais, M., Ali, U., Ali, T., Hamdi, M., Alghamdi, A. & Muhammad, F. (2020). Big data analytics for short and medium-term electricity load forecasting using an ai techniques ensembler. *Energies*, 13(19). https://doi.org/10.3390/en13195193.

Azadeh, A., Taghipour, M., Asadzadeh, S. M. & Abdollahi, M. (2014). Artificial immune simulation for improved forecasting of electricity consumption with random variations. *International Journal of Electrical Power & Energy Systems*, 55, 205–24. https://doi.org/10.1016/j.ijepes.2013.08.017.

Bala, P. K. (2010). Decision tree based demand forecasts for improving inventory performance. In 2010 IEEE International Conference on Industrial Engineering and Engineering Management, 1926–1930. https://doi.org/10.1109/IEEM.2010.5674628.

Barzegar, R., Aalami, M. T. & Adamowski, J. (2020). Short-term water quality variable prediction using a hybrid CNN–LSTM deep learning model. *Stochastic Environmental Research and Risk Assessment*, 34(2), 415–33. https://doi.org/10.1007/s00477-020-01776-2.

Bedi, J. & Toshniwal, D. (2018). Empirical mode decomposition based deep learning for electricity demand forecasting. *IEEE Access*, 6, 49144–56. https://doi.org/10.1109/ACCESS.2018.2867681.

Bejarano-Luque, J. L., Toril, M., Fernández-Navarro, M., Gijón, C. & Luna-Ramírez, S. (2021). A deep-learning model for estimating the impact of social events on traffic demand on a cell basis. *IEEE Access*, 9, 71673–86. https://doi.org/10.1109/ACCESS.2021.3078113.

Bennett, N. & Lemoine, G. J. (2014). What a difference a word makes: understanding threats to performance in a VUCA world." *Business Horizons*, 57(3), 311–17. https://doi.org/10.1016/j.bushor.2014.01.001.

Beyca, O. F., Ervural, B. C., Tatoglu, E., Ozuyar, P. G. & Zaim, S. (2019). Using machine learning tools for forecasting natural gas consumption in the province of Istanbul. *Energy Economics*, 80, 937–49. https://doi.org/10.1016/j.eneco.2019.03.006.

Bin, J. & Tianli, X. (2020). Forecast of export demand based on artificial neural network and fuzzy system theory."= *Journal of Intelligent & Fuzzy Systems*, 39(2), 1701–9. https://doi.org/10.3233/JIFS-179944.

Bolandnazar, E., Rohani, A. & Taki, M. (2020). Energy consumption forecasting in agriculture by artificial intelligence and mathematical models. *Energy Sources, Part A: Recovery, Utilization and Environmental Effects*, 42(13), 1618–1632. https://doi.org/10.1080/15567036.2019.1604872.

Bousqaoui, H., Achchab, S. & Tikito, K. (2017). Machine learning applications in supply chains: An emphasis on neural network applications." In 2017 3rd International Conference of Cloud Computing Technologies and Applications (CloudTech), 1–7. https://doi.org/10.1109/CloudTech.2017.8284722.

Bozkir, A. S. & Sezer, E. A. (2011). Predicting food demand in food courts by decision tree approaches. *Procedia Computer Science*, 3, 759–63. https://doi.org/10.1016/j.procs.2010.12.125.

Brereton, P., Kitchenham, B. A., Budgen, D., Turner, M. & Khalil, M. (2007). Lessons from applying the systematic literature review process within the software engineering domain. *Journal of Systems and Software*, Software Performance, 80(4), 571–83. https://doi.org/10.1016/j.jss.2006.07.009.

Burinskiene, A. (2018). Pharma supply chain: efficiency modelling approach," (2), 9.

Catal, C., Fenerci, A., Ozdemir, B. & Gulmez, O. (2015). Improvement of demand forecasting models with special days. *Procedia Computer Science*, 59, 262–67. https://doi.org/10.1016/j.procs.2015.07.554.

Centre for Reviews and Dissemination, University of York. 2001. undertaking systematic reviews of research on effectiveness: crd's guidance for carrying out or commissioning reviews. Monograph. York, UK: NHS Centre for Reviews and Dissemination. March 2001. http://www.york.ac.uk/inst/crd/crdreports.htm.

Čerka, P., Grigienė, J. & Sirbikytė, G. (2015). Liability for damages caused by artificial intelligence. *Computer Law & Security Review*, 31(3), 376–89. https://doi.org/10.1016/j.clsr.2015.03.008.

Chandra, P. & Fisher, M. L. (1994). Coordination of production and distribution planning. *European Journal of Operational Research*, 72(3), 503–17. https://doi.org/10.1016/0377-2217(94)90419-7.

Chandriah, K. K. & Naraganahalli, R. V. (2021). RNN/LSTM with modified adam optimizer in deep learning approach for automobile spare parts demand forecasting. *Multimedia Tools and Applications*, 80(17), 26145–59. https://doi.org/10.1007/s11042-021-10913-0.

Chang, P. -C., Fan, C. -Y. & Lin, J. -J. (2011). Monthly electricity demand forecasting based on a weighted evolving fuzzy neural network approach. *International Journal of Electrical Power & Energy Systems*, 33(1), 17–27. https://doi.org/10.1016/j.ijepes.2010.08.008.

Chaturvedi, D. K., A. P. Sinha, and O. P. Malik. 2015. "Short Term Load Forecast Using Fuzzy Logic and Wavelet Transform Integrated Generalized Neural Network." *International Journal of Electrical Power & Energy Systems* 67: 230–37. https://doi.org/10.1016/j.ijepes.2014.11.027.

Chauhan, Bhavesh Kumar, and Madasu Hanmandlu. (2010). Load Forecasting Using Wavelet Fuzzy Neural Network. *International Journal of Knowledge-Based and Intelligent Engineering Systems*, 14(2), 57–71. https://doi.org/10.3233/KES-2010-0192.

Chen, Albert Y., Tsung-Yu Lu, Matthew Huei-Ming Ma, and Wei-Zen Sun. (2016). Demand Forecast Using Data Analytics for the Preallocation of Ambulances. *IEEE Journal of Biomedical and Health Informatics*, 20(4), 1178–87. https://doi.org/10.1109/JBHI.2015.2443799.

Chen, Argon, C.-H. Hsu, and J. Blue. (2007). Demand Planning Approaches to Aggregating and Forecasting Interrelated Demands for Safety Stock and Backup Capacity Planning. *International Journal of Production Research*, 45(10), 2269–94. https://doi.org/10.1080/00207540600690693.

Chen, Chun-Fu, Ming-Cheng Lai, and Ching-Chiang Yeh. (2012). ForecastingTourism Demand Based on Empirical Mode Decomposition and Neural Network.Knowledge-BasedSystems,26,281–87.https://doi.org/10.1016/j.knosys.2011.09.002.

Chen, G., T. Long, Y. Bai, and J. Zhang. (2019). A Forecasting Framework Based on Kalman Filter Integrated Multivariate Local Polynomial Regression: Application to Urban Water Demand. *Neural Processing Letters*, 50(1), 497–513. https://doi.org/10.1007/s11063-019-10001-3.

Chen, Jeng-Fung, Shih-Kuei Lo, and Quang Hung Do. (2017). Forecasting Monthly Electricity Demands: An Application of Neural Networks Trained by Heuristic Algorithms. *Information*, 8(1), 31. https://doi.org/10.3390/info8010031.

Chen, Li-Yueh. (2014). Application of SVR with Chaotic GASA Algorithm to Forecast Taiwanese 3G Mobile Phone Demand. *Neurocomputing*, 127, 206–13. https://doi.org/10.1016/j.neucom.2013.08.010.

Chen, Toly, and Yu-Cheng Wang. (2012). Long-Term Load Forecasting by a Collaborative Fuzzy-Neural Approach. *International Journal of Electrical Power & Energy Systems*, 43(1), 454–64. https://doi.org/10.1016/j.ijepes.2012.05.072.

Cheng, C.-H., and L.-Y. Wei. (2010). One Step-Ahead ANFIS Time Series Model for Forecasting Electricity Loads. *Optimization and Engineering*, 11(2), 303–17. https://doi.org/10.1007/s11081-009-9091-5.

Choi, Tsan-Ming, Chi-Leung Hui, Na Liu, Sau-Fun Ng, and Yong Yu. (2014). Fast Fashion Sales Forecasting with Limited Data and Time. *Decision Support Systems*, 59, 84–92. https://doi.org/10.1016/j.dss.2013.10.008.

Chong, Eunsuk, Chulwoo Han, and Frank C. Park. (2017). Deep Learning Networks for Stock Market Analysis and Prediction: Methodology, Data Representations, and Case Studies. *Expert Systems with Applications*, 83 (October), 187–205. https://doi.org/10.1016/j.eswa.2017.04.030.

Chopra, Sunil, and Peter Meindl. (2007). Supply Chain Management. Strategy, Planning & Operation. In *Das Summa Summarum des Management: Die 25 wichtigsten Werke für Strategie, Führung und Veränderung*, edited by Cornelius Boersch and Rainer Elschen, 265–75. Wiesbaden: Gabler. https://doi.org/10.1007/978-3-8349-9320-5_22.

Christopher, Martin, and Matthias Holweg. (2017). Supply Chain 2.0 Revisited: A Framework for Managing Volatility-Induced Risk in the Supply Chain. *International Journal of Physical Distribution & Logistics Management*, 47(1), 2–17. https://doi.org/10.1108/IJPDLM-09-2016-0245.

Chu, F. -L. (2014). Using a Logistic Growth Regression Model to Forecast the Demand for Tourism in Las Vegas. *Tourism Management Perspectives*, 12, 62–67. https://doi.org/10.1016/j.tmp.2014.08.003.

Ciulla, G. & A. D'Amico. (2019). Building Energy Performance Forecasting: A Multiple Linear Regression Approach. *Applied Energy*, 253. https://doi.org/10.1016/j.apenergy.2019.113500.

Clarke, S. (2006). Transformation Lessons from Coca-Cola Enterprises Inc.: Managing the Introduction of a Structured Forecast Process. *Foresight: The International Journal of Applied Forecasting*, (4), 21–25.

Colombet, Isabelle. (2015). Revue systématique et méta-analyse en médecine palliative. *Médecine Palliative : Soins de Support - Accompagnement - Éthique*, 14(4), 240–53. https://doi.org/10.1016/j.medpal.2015.04.001.

Daniel, L. Marino, Amarasinghe Kasun, & Manic Milos. (2016). Building Energy Load Forecasting Using Deep Neural Networks. 2016. https://proxy.univh2c.ma:2291/document/7793413/. Das, Gulesin Sena. (2017). Forecasting the Energy Demand of Turkey with a NN Based on an Improved Particle Swarm Optimization. *Neural Computing & Applications* 28 (December), \$539–49. https://doi.org/10.1007/s00521-016-2367-8.

Dave, Emmanuel, Albert Leonardo, Marethia Jeanice, & Novita Hanafiah. (2021). Forecasting Indonesia Exports Using a Hybrid Model ARIMA-LSTM. *Procedia Computer Science*, 5th International Conference on Computer Science and Computational Intelligence 2020, 179 (January), 480–87. https://doi.org/10.1016/j.procs.2021.01.031.

Delorme-Costil, Alexandra, & Jean-Jacques Bezian. (2017). Forecasting Domestic Hot Water Demand in Residential House Using Artificial Neural Networks. In 2017 16th IEEE International Conference on Machine Learning and Applications (ICMLA), 467–72. https://doi.org/10.1109/ICMLA.2017.0-117.

Demiriz, Ayhan. (2014). Demand Forecasting Based on Pairwise Item Associations. *Procedia Computer Science*, 36, 261–68. https://doi.org/10.1016/j.procs.2014.09.091.

Domingos, E., B. Ojeme, & O. Daramola. (2021). Experimental Analysis of Hyperparameters for Deep Learning-based Churn Prediction in the Banking Sector. *Computation*, 9(3). https://doi.org/10.3390/computation9030034.

Du, Wei, Sunney Yung Sun Leung, & Chun Kit Kwong. (2015). A Multiobjective Optimization-Based Neural Network Model for Short-Term Replenishment Forecasting in Fashion Industry. *Neurocomputing*, 151, 342–53. https://doi.org/10.1016/j.neucom.2014.09.030.

Dwivedi, Yogesh K., Laurie Hughes, Elvira Ismagilova, Gert Aarts, Crispin Coombs, Tom Crick, Yanqing Duan. (2021). Artificial Intelligence (AI): Multidisciplinary Perspectives on Emerging Challenges, Opportunities, and Agenda for Research, Practice and Policy. *International Journal of Information Management* 57 (April), 101994. https://doi.org/10.1016/j.ijinfomgt.2019.08.002.

EL Filali, Aicha, El Habib Ben Lahmer, & Sanaa EL ELFilali. (2021). Exploring Applications of Machine Learning for Supply Chain Management. In , 46–52. IEEE Computer Society. https://doi.org/10.1109/TST52996.2021.00015.

Eseye, A. T. & M. Lehtonen. (2020). Short-Term Forecasting of Heat Demand of Buildings for Efficient and Optimal Energy Management Based on Integrated Machine Learning Models. *IEEE Transactions on Industrial Informatics*, 16(12), 7743–55. https://doi.org/10.1109/TII.2020.2970165.

Fan, Guo-Feng, Yan-Hui Guo, Jia-Mei Zheng, & Wei-Chiang Hong. (2020). A Generalized Regression Model Based on Hybrid Empirical Mode Decomposition and Support Vector Regression with Back-Propagation Neural Network for Mid-Short-

Term Load Forecasting. *Journal of Forecasting*, 39(5), 737–56. https://doi.org/10.1002/for.2655.

Fan, Guo-Feng, Xiang-Ru Jin, & Wei-Chiang Hong. (2021). Application of COEMD-S-SVR Model in Tourism Demand Forecasting and Economic Behavior Analysis: The Case of Sanya City. *Journal of the Operational Research Society*, April. https://doi.org/10.1080/01605682.2021.1915192.

Fatorachian, Hajar, & Hadi Kazemi. (2018). A Critical Investigation of Industry 4.0 in Manufacturing: Theoretical Operationalisation Framework. *Production Planning & Control*, 29(8), 33–44. https://doi.org/10.1080/09537287.2018.1424960.

Feizabadi, Javad. 2020. "Machine Learning Demand Forecasting and Supply Chain Performance." *International Journal of Logistics Research and Applications*, August, 1–24. https://doi.org/10.1080/13675567.2020.1803246.

Fleischmann, Bernhard, Herbert Meyr, & Michael Wagner. (2002). Advanced Planning. In *Supply Chain Management and Advanced Planning: Concepts, Models, Software and Case Studies*, edited by Hartmut Stadtler and Christoph Kilger, 71–96. Berlin, Heidelberg: Springer. https://doi.org/10.1007/978-3-662-10142-1_5.

Fradinata, E., S. Suthummanon, W. Suntiamorntut, & M.M. Noor. (2019). Compare the Forecasting Method of Artificial Neural Network and Support Vector Regression Model to Measure the Bullwhip Effect in Supply Chain. *Journal of Mechanical Engineering* and Sciences, 13(2), 4816–34. https://doi.org/10.15282/jmes.13.2.2019.04.0401.

Francis, Hudson, and Andrew Kusiak. (2017). Prediction of Engine Demand with a Data-Driven Approach. *Procedia Computer Science*, XII International Symposium Intelligent Systems 2016, INTELS 2016, 5-7 October 2016, Moscow, Russia, 103 (January): 28–35. https://doi.org/10.1016/j.procs.2017.01.005.

Fu, W. & C. -F. Chien. (2019). UNISON Data-Driven Intermittent Demand Forecast Framework to Empower Supply Chain Resilience and an Empirical Study in Electronics Distribution. *Computers and Industrial Engineering*, 135, 940–49. https://doi.org/10.1016/j.cie.2019.07.002.

Fu, Y., Z. Li, H. Zhang, & P. Xu. (2015). Using Support Vector Machine to Predict Next Day Electricity Load of Public Buildings with Sub-Metering Devices. In, 121, 1016–22. https://doi.org/10.1016/j.proeng.2015.09.097.

Gao, X. & G.M. Lee. (2019). Moment-Based Rental Prediction for Bicycle-Sharing Transportation Systems Using a Hybrid Genetic Algorithm and Machine Learning. *Computers and Industrial Engineering*, 128, 60–69. https://doi.org/10.1016/j.cie.2018.12.023. Germi, Masoud Bakhshi, Mohammad Mirjavadi, Aghil Seyed Sadeghi Namin, & Aliasghar Baziar. (2014). A Hybrid Model for Daily Peak Load Power Forecasting Based on SAMBA and Neural Network. *Journal of Intelligent & Fuzzy Systems*, 27(2), 913–920. https://doi.org/10.3233/IFS-131049.

Ghanbari, Arash, S. M. R. Kazemi, Farhad Mehmanpazir, & Mohammad Masoud Nakhostin. (2013). A Cooperative Ant Colony Optimization-Genetic Algorithm Approach for Construction of Energy Demand Forecasting Knowledge-Based Expert Systems. *Knowledge-Based Systems*, 39, 194–206. https://doi.org/10.1016/j.knosys.2012.10.017.

Gonçalves, João N. C., Paulo Cortez, M. Sameiro Carvalho, & Nuno M. Frazão. (2021). A Multivariate Approach for Multi-Step Demand Forecasting in Assembly Industries: Empirical Evidence from an Automotive Supply Chain. *Decision Support Systems*, 142 (March), 113452. https://doi.org/10.1016/j.dss.2020.113452.

Goodfellow, Ian, Yoshua Bengio, & Aaron Courville. (2016). *Deep Learning*. Adaptive Computation and Machine Learning Series. Cambridge, MA, USA: MIT Press.

Gunasekaran, A, C Patel, & Ronald E McGaughey. (2004). A Framework for Supply Chain Performance Measurement. *International Journal of Production Economics*, Supply Chain Management for the 21st Century Organizational Competitiveness, 87(3), 333–47. https://doi.org/10.1016/j.ijpe.2003.08.003.

Guo, Hongpeng, Cheng Guo, Beichun Xu, Yujie Xia, & Fanhui Sun. (2021). MLP Neural Network-Based Regional Logistics Demand Prediction. *Neural Computing & Applications*, 33(9), 3939–3952. https://doi.org/10.1007/s00521-020-05488-0.

Haberleitner, Helmut, Herbert Meyr, & Alfred Taudes. (2010). Implementation of a Demand Planning System Using Advance Order Information. *International Journal of Production Economics*, Supply Chain Forecasting Systems, 128(2), 518–26. https://doi.org/10.1016/j.ijpe.2010.07.003.

Hamid, M. B. Abdul, & T. K. Abdul Rahman. (2010). Short Term Load Forecasting Using an Artificial Neural Network Trained by Artificial Immune System Learning Algorithm. In 2010 12th International Conference on Computer Modelling and Simulation, 408–413. https://doi.org/10.1109/UKSIM.2010.82.

Hamzacebi, Coskun, Huseyin Es Avni, & Recep Cakmak. (2019). Forecasting of Turkey's Monthly Electricity Demand by Seasonal Artificial Neural Network. *Neural Computing & Applications*, 31(7), 2217–2231. https://doi.org/10.1007/s00521-017-3183-5.

Haq, E. U., X. Lyu, Y. Jia, M. Hua, & F. Ahmad. (2020). fForecasting Household Electric Appliances Consumption and Peak Demand Based on Hybrid Machine

Learning Approach." *Energy Reports*, 6, 1099–1105. https://doi.org/10.1016/j.egyr.2020.11.071.

Haque, Ashraf Ul, Paras Mandal, Julian Meng, & Ricardo L. Pineda. (2012). "Performance Evaluation of Different Optimization Algorithms for Power Demand Forecasting Applications in a Smart Grid Environment. *Procedia Computer Science*, 12, 320–325. https://doi.org/10.1016/j.procs.2012.09.078.

He, Wan. (2017). Load Forecasting via Deep Neural Networks. *Procedia Computer Science*, 5th International Conference on Information Technology and Quantitative Management, ITQM 2017, 122 (January), 308–14. https://doi.org/10.1016/j.procs.2017.11.374.

Herrera, M., L. Torgo, J. Izquierdo, & R. Pérez-García. (2010). Predictive Models for Forecasting Hourly Urban Water Demand. *Journal of Hydrology*, 387(1–2), 141–50. https://doi.org/10.1016/j.jhydrol.2010.04.005.

Hong, Tao, Min Gui, Mesut E. Baran, & H. Lee Willis. (2010). Modeling and Forecasting Hourly Electric Load by Multiple Linear Regression with Interactions. In *IEEE PES General Meeting*, 1–8. https://doi.org/10.1109/PES.2010.5589959.

Hong, Wei-Chiang, Yucheng Dong, Li-Yueh Chen, & Shih-Yung Wei. (2011). SVR with Hybrid Chaotic Genetic Algorithms for Tourism Demand Forecasting. *Applied Soft Computing*, 11(2), 1881–90. https://doi.org/10.1016/j.asoc.2010.06.003.

Hong, Wei-Chiang, Yucheng Dong, Wen Yu Zhang, Li-Yueh Chen, & B. K. Panigrahi. (2013). Cyclic Electric Load Forecasting by Seasonal SVR with Chaotic Genetic Algorithm. *International Journal of Electrical Power & Energy Systems*, 44(1), 604–614. https://doi.org/10.1016/j.ijepes.2012.08.010.

Hošovský, A., J. Pitel', M. Adámek, J. Mižáková, & K. Židek. (2021). Comparative Study of Week-Ahead Forecasting of Daily Gas Consumption in Buildings Using Regression ARMA/SARMA and Genetic-Algorithm-Optimized Regression Wavelet Neural Network Models. *Journal of Building Engineering*, 34. https://doi.org/10.1016/j.jobe.2020.101955.

Hossen, Tareq, Arun Sukumaran Nair, Radhakrishnan Angamuthu Chinnathambi, and Prakash Ranganathan. 2018. "Residential Load Forecasting Using Deep Neural Networks (DNN)." In 2018 North American Power Symposium (NAPS), 1–5. https://doi.org/10.1109/NAPS.2018.8600549.

Hribar, Rok, Primož Potočnik, Jurij Šilc, & Gregor Papa. (2019). A comparison of models for forecasting the residential natural gas demand of an urban area. *Energy*, 167(C), 511–22.

Hu, Yi-Chung. (2020). Constructing Grey Prediction Models Using Grey Relational Analysis and Neural Networks for Magnesium Material Demand Forecasting. *Applied Soft Computing*, 93 (August), 106398. https://doi.org/10.1016/j.asoc.2020.106398.

Hu, Yi-Chung, Peng Jiang, & Ping-Chuan Lee. (2019). Forecasting Tourism Demand by Incorporating Neural Networks into Grey–Markov Models. *Journal of the Operational Research Society*, 70(1), 12–20. https://doi.org/10.1080/01605682.2017.1418150.

Hu, Zhongyi, Yukun Bao, & Tao Xiong. (2014). Comprehensive Learning Particle Swarm Optimization Based Memetic Algorithm for Model Selection in Short-Term Load Forecasting Using Support Vector Regression. *Applied Soft Computing*, 25, 15–25. https://doi.org/10.1016/j.asoc.2014.09.007.

Huang, Lijuan, Guojie Xie, Wende Zhao, Yan Gu, & Yi Huang. (2021). Regional Logistics Demand Forecasting: A BP Neural Network Approach. *Complex & Intelligent Systems*, March. https://doi.org/10.1007/s40747-021-00297-x.

Huang, Y., Y. Yuan, H. Chen, J. Wang, Y. Guo, & T. Ahmad. (2019). A Novel Energy Demand Prediction Strategy for Residential Buildings Based on Ensemble Learning. In , 158:3411–16. https://doi.org/10.1016/j.egypro.2019.01.935.

Huang, Zihao, Gang Huang, Zhijun Chen, Chaozhong Wu, Xiaofeng Ma, & Haobo Wang. (2019). Multi-Regional Online Car-Hailing Order Quantity Forecasting Based on the Convolutional Neural Network. *Information*, 10(6), 193. https://doi.org/10.3390/info10060193.

Iansiti, Marco, & Karim R. Lakhani. (2020). *Competing in the Age of AI: Strategy and Leadership When Algorithms and Networks Run the World*. Boston, MA: Harvard Business Review Press.

Idowu, S., S. Saguna, C. Åhlund, and O. Schelén. (2016). Applied Machine Learning: Forecasting Heat Load in District Heating System. *Energy and Buildings*, 133, 478– 88. https://doi.org/10.1016/j.enbuild.2016.09.068.

Ivanov, Dmitry, and Maxim Rozhkov. (2020). Coordination of Production and Ordering Policies under Capacity Disruption and Product Write-off Risk: An Analytical Study with Real-Data Based Simulations of a Fast Moving Consumer Goods Company. *Annals of Operations Research*, 291(1), 387–407. https://doi.org/10.1007/s10479-017-2643-8.

Jaipuria, Sanjita, and S. S. Mahapatra. (2014). An Improved Demand Forecasting Method to Reduce Bullwhip Effect in Supply Chains. *Expert Systems with Applications*, 41(5), 2395–2408. https://doi.org/10.1016/j.eswa.2013.09.038.

Jaramillo-Morán, Miguel A., Eva González-Romera, and Diego Carmona-Fernández. (2013). Monthly Electric Demand Forecasting with Neural Filters. *International Journal of Electrical Power & Energy Systems*, 49, 253–63. https://doi.org/10.1016/j.ijepes.2013.01.019.

Jawad, M., M.S.A. Nadeem, S.-O. Shim, I.R. Khan, A. Shaheen, N. Habib, L. Hussain, & W. Aziz. (2020). Machine Learning Based Cost Effective Electricity Load Forecasting Model Using Correlated Meteorological Parameters. *IEEE Access*, 8, 146847–64. https://doi.org/10.1109/ACCESS.2020.3014086.

Jeble, Shirish, Sneha Kumari, & Yogesh Patil. (2017). Role of Big Data in Decision Making. *Operations and Supply Chain Management: An International Journal*, 11(1), 36–44. https://doi.org/10.31387/oscm0300198.

Ji, S., X. Wang, W. Zhao, & D. Guo. (2019). An Application of a Three-Stage XGboost-Based Model to Sales Forecasting of a Cross-Border e-Commerce Enterprise. *Mathematical Problems in Engineering*. https://doi.org/10.1155/2019/8503252.

Jin, F., Y. Li, S. Sun, & H. Li. (2020). Forecasting Air Passenger Demand with a New Hybrid Ensemble Approach. *Journal of Air Transport Management*, 83. https://doi.org/10.1016/j.jairtraman.2019.101744.

Johannesen, N.J., M. Kolhe, & M. Goodwin. (2019). Relative Evaluation of Regression Tools for Urban Area Electrical Energy Demand Forecasting. *Journal of Cleaner Production*, 218, 555–564. https://doi.org/10.1016/j.jclepro.2019.01.108.

Kanavos, Andreas, Fotios Kounelis, Lazaros Iliadis, & Christos Makris. (2021). Deep Learning Models for Forecasting Aviation Demand Time Series. *Neural Computing & Applications*, June. https://doi.org/10.1007/s00521-021-06232-y.

Kang, Zilu, Yuting Zuo, Zhibin Huang, Feng Zhou, & Penghui Chen. (2017). Research on the Forecast of Shared Bicycle Rental Demand Based on Spark Machine Learning Framework. In 2017 16th International Symposium on Distributed Computing and Applications to Business, Engineering and Science (DCABES), 219– 222. https://doi.org/10.1109/DCABES.2017.55.

Kankal, Murat, and Ergun Uzlu. 2017. "Neural Network Approach with Teaching-Learning-Based Optimization for Modeling and Forecasting Long-Term Electric Energy Demand in Turkey." *Neural Computing & Applications* 28 (December): S737–47. https://doi.org/10.1007/s00521-016-2409-2.

Kantasa-ard, Anirut, Maroua Nouiri, Abdelghani Bekrar, Abdessamad Ait el cadi, & Yves Sallez. 2020. Machine Learning for Demand Forecasting in the Physical Internet: A Case Study of Agricultural Products in Thailand. *International Journal of Production Research*, 0(0), 1–25. https://doi.org/10.1080/00207543.2020.1844332.

Kaytez, Fazil, M. Cengiz Taplamacioglu, Ertugrul Cam, & Firat Hardalac. (2015). Forecasting Electricity Consumption: A Comparison of Regression Analysis, Neural Networks and Least Squares Support Vector Machines. *International Journal of Electrical Power & Energy Systems*, 67, 431–438. https://doi.org/10.1016/j.ijepes.2014.12.036.

Ke, J., H. Zheng, H. Yang, and X.M. Chen. (2017). Short-Term Forecasting of Passenger Demand under on-Demand Ride Services: A Spatio-Temporal Deep Learning Approach. *Transportation Research Part C: Emerging Technologies*, 85, 591–608. https://doi.org/10.1016/j.trc.2017.10.016.

Ke, Jintao, Hongyu Zheng, Hai Yang, & Xiqun (Michael) Chen. (2017). Short-Term Forecasting of Passenger Demand under on-Demand Ride Services: A Spatio-Temporal Deep Learning Approach. *Transportation Research Part C: Emerging Technologies* 85 (December), 591–608. https://doi.org/10.1016/j.trc.2017.10.016.

Kilimci, Zeynep Hilal, A. Okay Akyuz, Mitat Uysal, Selim Akyokus, M. Ozan Uysal, Berna Atak Bulbul, & Mehmet Ali Ekmis. (2019). An Improved Demand Forecasting Model Using Deep Learning Approach and Proposed Decision Integration Strategy for Supply Chain. *Complexity* 2019 (March), 1–15. https://doi.org/10.1155/2019/9067367.

Kim, Jaedong, Hanjun Lee, & Soobin Choi. 2018. Machine Learning Based Approach for Demand Forecasting Anti-Aircraft Missiles. In 2018 5th International Conference on Industrial Engineering and Applications (ICIEA), 367–72. https://doi.org/10.1109/IEA.2018.8387126.

Kim, Jeong beom. (2019). Implementation of Artificial Intelligence System and Traditional System: A Comparative Study. *Journal of System and Management Sciences*, September. https://doi.org/10.33168/JSMS.2019.0309.

Kim, Myoung-Jong. (2012). Ensemble Learning with Support Vector Machines for Bond Rating. *Journal of Intelligence and Information Systems*, 18(2), 29–45. https://doi.org/10.13088/jiis.2012.18.2.029.

Kitchenham, B. (2004). Procedures for Performing Systematic Reviews. *Undefined*. https://www.semanticscholar.org/paper/Procedures-for-Performing-Systematic-Reviews-Kitchenham/29890a936639862f45cb9a987dd599dce9759bf5.

Kone, Estelle R. S., & Mark H. Karwan. (2011). Combining a New Data Classification Technique and Regression Analysis to Predict the Cost-To-Serve New Customers. *Computers & Industrial Engineering*, 61(1), 184–97. https://doi.org/10.1016/j.cie.2011.03.009.

Kotillova, A., I. Koprinska, & M. Rana. 2012. Statistical and Machine Learning Methods for Electricity Demand Prediction. *Lecture Notes in Computer Science*

(Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics) 7664 LNCS (PART 2), 535–42. https://doi.org/10.1007/978-3-642-34481-7_65.

Kumar, Ajay, Ravi Shankar, & Naif Radi Aljohani. (2020). A Big Data Driven Framework for Demand-Driven Forecasting with Effects of Marketing-Mix Variables. *Industrial Marketing Management*, 90 (October), 493–507. https://doi.org/10.1016/j.indmarman.2019.05.003.

Kumar, J. & A.K. Singh. (2020). Decomposition Based Cloud Resource Demand Prediction Using Extreme Learning Machines. *Journal of Network and Systems Management*, 28(4), 1775–93. https://doi.org/10.1007/s10922-020-09557-6.

Laaroussi, Houria, Fatima Guerouate, & Mohamed sbihi. (2020). Deep Learning Framework for Forecasting Tourism Demand. In 2020 IEEE International Conference on Technology Management, Operations and Decisions (ICTMOD), 1–4. https://doi.org/10.1109/ICTMOD49425.2020.9380612.

Lahouar, A. & J. Ben Hadj Slama. (2015). Random Forests Model for One Day Ahead Load Forecasting. In . https://doi.org/10.1109/IREC.2015.7110975.

Laik, M. N., M. Choy, & P. Sen. (2014). Predicting Airline Passenger Load: A Case Study. In , 1, 33–38. https://doi.org/10.1109/CBI.2014.39.

Lau, H. C. W., G. T. S. Ho, & Yi Zhao. (2013). A Demand Forecast Model Using a Combination of Surrogate Data Analysis and Optimal Neural Network Approach. *Decision Support Systems*, 54(3), 1404–16. https://doi.org/10.1016/j.dss.2012.12.008.

Lee, H., S.G. Kim, H.-W. Park, and P. Kang. 2014. "Pre-Launch New Product Demand Forecasting Using the Bass Model: A Statistical and Machine Learning-Based Approach." *Technological Forecasting and Social Change* 86: 49–64. https://doi.org/10.1016/j.techfore.2013.08.020.

Lee, Hau L., V. Padmanabhan, and Seungjin Whang. 1997. "Information Distortion in a Supply Chain: The Bullwhip Effect." *Management Science* 43 (4): 546–58. https://doi.org/10.1287/mnsc.43.4.546.

Li, Hong-ze, Sen Guo, Chun-jie Li, & Jing-qi Sun. (2013). A Hybrid Annual Power Load Forecasting Model Based on Generalized Regression Neural Network with Fruit Fly Optimization Algorithm. *Knowledge-Based Systems*, 37, 378–87. https://doi.org/10.1016/j.knosys.2012.08.015.

Li, M. S., J. L. Wu, T. Y. Ji, Q. H. Wu, & L. Zhu. 2015. Short-Term Load Forecasting Using Support Vector Regression-Based Local Predictor. In 2015 IEEE Power Energy Society General Meeting, 1–5. https://doi.org/10.1109/PESGM.2015.7285911. Li, Penghua, Yinguo Li, Qingyu Xiong, Yi Chai, & Yi Zhang. 2014. Application of a Hybrid Quantized Elman Neural Network in Short-Term Load Forecasting. *International Journal of Electrical Power & Energy Systems*, 55, 749–59. https://doi.org/10.1016/j.ijepes.2013.10.020.

Li, Xuefeng, Yong Zhang, Mingyang Du, & Jingzong Yang. (2020). The Forecasting of Passenger Demand under Hybrid Ridesharing Service Modes: A Combined Model Based on WT-FCBF-LSTM. *Sustainable Cities and Society* 62 (November): 102419. https://doi.org/10.1016/j.scs.2020.102419.

Liu, Chang, Zhijian Jin, Jie Gu, & Caiming Qiu. (2017). Short-Term Load Forecasting Using a Long Short-Term Memory Network. In 2017 IEEE PES Innovative Smart Grid Technologies Conference Europe (ISGT-Europe), 1–6. https://doi.org/10.1109/ISGTEurope.2017.8260110.

Liu, J., L. Lin, Z. Li, H. Guo, & Y. Lv. 2020. Spare Aeroengine Demand Prediction Model Based on Deep Croston Method. *Journal of Aerospace Information Systems*, 17(2), 125–33. https://doi.org/10.2514/1.I010707.

Liu, P., W. Ming, & B. Hu. (2021). Sales Forecasting in Rapid Market Changes Using a Minimum Description Length Neural Network. *Neural Computing and Applications*, 33(3), 937–48. https://doi.org/10.1007/s00521-020-05294-8.

Livieris, Ioannis E., Emmanuel Pintelas, Theodore Kotsilieris, Stavros Stavroyiannis, & Panagiotis Pintelas. (2019). Weight-Constrained Neural Networks in Forecasting Tourist Volumes: A Case Study. *Electronics*, 8(9), 1005. https://doi.org/10.3390/electronics8091005.

Lou, Chin Wang, & Ming Chui Dong. (2015). A Novel Random Fuzzy Neural Networks for Tackling Uncertainties of Electric Load Forecasting. *International Journal of Electrical Power & Energy Systems*, 73, 34–44. https://doi.org/10.1016/j.ijepes.2015.03.003.

Lu, Chi-Jie. (2014). Sales Forecasting of Computer Products Based on Variable Selection Scheme and Support Vector Regression. *Neurocomputing*, 128, 491–99. https://doi.org/10.1016/j.neucom.2013.08.012.

Lu, Chi-Jie, Tian-Shyug Lee, & Chia-Mei Lian. (2012). Sales Forecasting for Computer Wholesalers: A Comparison of Multivariate Adaptive Regression Splines and Artificial Neural Networks. *Decision Support Systems*, 54(1), 584–96. https://doi.org/10.1016/j.dss.2012.08.006.

Ma, Hongjiang, and Xu Luo. (2021). Logistics Demand Forecasting Model Based on Improved Neural Network Algorithm. *Journal of Intelligent & Fuzzy Systems*, 40(4), 6385–95. https://doi.org/10.3233/JIFS-189479.

Mahmoud, Z. (2021). A new sales forecasting method for industrial supply chain. *International Journal of Smart Business and Technology*, 9(2), 1-12. https://doi.org/10.21742/IJSBT.2021.9.2.01

Marinescu, Andrei, Ivana Dusparic, Colin Harris, Vinny Cahill, & Siobhän Clarke. 2014. A Dynamic Forecasting Method for Small Scale Residential Electrical Demand. In 2014 International Joint Conference on Neural Networks (IJCNN), 3767–74. https://doi.org/10.1109/IJCNN.2014.6889425.

Mason-Jones, Rachel, & Denis R. Towill. (1999). Vaincre l'incertitude : Réduire l'effet Coup de Fouet Sur Les Supply Chains. *Logistique & Management*, 7(1), 7–14. https://doi.org/10.1080/12507970.1999.11516693.

Mccarthy, Teresa M., Donna F. Davis, Susan L. Golicic, & John T. Mentzer. (2006). The Evolution of Sales Forecasting Management: A 20-Year Longitudinal Study of Forecasting Practices. *Journal of Forecasting*, 25(5), 303–324. https://doi.org/10.1002/for.989.

Merkuryeva, Galina, Aija Valberga, & Alexander Smirnov. (2019). Demand Forecasting in Pharmaceutical Supply Chains: A Case Study. *Procedia Computer Science*, ICTE in Transportation and Logistics 2018 (ICTE 2018), 149 (January), 3–10. https://doi.org/10.1016/j.procs.2019.01.100.

Milgrom, Paul, & John Roberts. (1988) Communication and Inventory as Substitutes in Organizing Production. *The Scandinavian Journal of Economics*, 90(3), 275–289. https://doi.org/10.2307/3440309.

Min, Hokey. (2010). Artificial Intelligence in Supply Chain Management: Theory and Applications. *International Journal of Logistics Research and Applications*, 13(1), 13–39. https://doi.org/10.1080/13675560902736537.

Minis, I, & Nicholas Ampazis. (2006). Applications of Neural Networks in Supply Chain Management. In , 589–607.

Mohammadi, Reza, S. M. T. Fatemi Ghomi, & Farzad Zeinali. 2014. A New Hybrid Evolutionary Based RBF Networks Method for Forecasting Time Series: A Case Study of Forecasting Emergency Supply Demand Time Series. *Engineering Applications of Artificial Intelligence*, 36, 204–14. https://doi.org/10.1016/j.engappai.2014.07.022.

Moon, Mark A., John T. Mentzer, & Carlo D. Smith. (2003). Conducting a Sales Forecasting Audit. *International Journal of Forecasting*, 19(1), 5–25. https://doi.org/10.1016/S0169-2070(02)00032-8.

Moroff, Nikolas Ulrich, Ersin Kurt, & Josef Kamphues. (2021). Machine Learning and Statistics: A Study for Assessing Innovative Demand Forecasting Models.

Procedia Computer Science, Proceedings of the 2nd International Conference on Industry 4.0 and Smart Manufacturing (ISM 2020), 180 (January), 40–49. https://doi.org/10.1016/j.procs.2021.01.127.

Nasser, A.A., M.Z. Rashad, & S.E. Hussein. (2020). A Two-Layer Water DemandPrediction System in Urban Areas Based on Micro-Services and LSTM NeuralNetworks.IEEEAccess,8,147647–61.https://doi.org/10.1109/ACCESS.2020.3015655.

Ni, Du, Zhi Xiao, & Ming K Lim. (2020). A Systematic Review of the Research Trends of Machine Learning in Supply Chain Management. *International Journal of Machine Learning and Cybernetics*, July. https://doi.org/10.1007/s13042-019-01050-0.

Niu, Dong-xiao, Hui-feng Shi, & Desheng Dash Wu. (2012). Short-Term Load Forecasting Using Bayesian Neural Networks Learned by Hybrid Monte Carlo Algorithm. *Applied Soft Computing*, 12(6), 1822–27. https://doi.org/10.1016/j.asoc.2011.07.001.

Nuaimi, N.A. 2014. Data Mining Approaches for Predicting Demand for Healthcare Services in Abu Dhabi. In, 42–47. https://doi.org/10.1109/INNOVATIONS.2014.6987559.

Ofori-Ntow Jnr, E., Y.Y. Ziggah, & S. Relvas. (2021). Hybrid Ensemble Intelligent Model Based on Wavelet Transform, Swarm Intelligence and Artificial Neural Network for Electricity Demand Forecasting. *Sustainable Cities and Society*, 66. https://doi.org/10.1016/j.scs.2020.102679.

Oke, Sunday. (2004). Artificial intelligence: a review of the literature. *Proceedings* of the Fifth Asia Pacific Industrial Engineering and Management Systems Conference, January.

Oliva, Rogelio, & Noel Watson. 2009. Managing Functional Biases in Organizational Forecasts: A Case Study of Consensus Forecasting in Supply Chain Planning. *Production and Operations Management*, 18(2), 138–51. https://doi.org/10.1111/j.1937-5956.2009.01003.x.

Olmedo, Elena. (2016). Comparison of Near Neighbour and Neural Network in Travel Forecasting. *Journal of Forecasting*, 35(3), 217–23. https://doi.org/10.1002/for.2370.

Packowski, Josef. (2013). *LEAN Supply Chain Planning: The New Supply Chain Management Paradigm for Process Industries to Master Today's VUCA World.* CRC Press.

Panagopoulos, A.A., F. Christianos, M. Katsigiannis, K. Mykoniatis, M. Pritoni, O.P. Panagopoulos, T. Peffer. (2020). A Low-Complexity Non-Intrusive Approach to Predict the Energy Demand of Buildings over Short-Term Horizons. *Advances in Building Energy Research*. https://doi.org/10.1080/17512549.2020.1835712.

Prado, F., M.C. Minutolo, & W. Kristjanpoller. (2020). Forecasting Based on an Ensemble Autoregressive Moving Average - Adaptive Neuro - Fuzzy Inference System – Neural Network - Genetic Algorithm Framework. *Energy*, 197. https://doi.org/10.1016/j.energy.2020.117159.

Priyadarshi, R., A. Panigrahi, S. Routroy, & G.K. Garg. 2019. Demand Forecasting at Retail Stage for Selected Vegetables: A Performance Analysis. *Journal of Modelling in Management*, 14(4), 1042–63. https://doi.org/10.1108/JM2-11-2018-0192.

Punia, S., K. Nikolopoulos, S.P. Singh, J.K. Madaan, & K. Litsiou. (2020). Deep Learning with Long Short-Term Memory Networks and Random Forests for Demand Forecasting in Multi-Channel Retail. *International Journal of Production Research*, 58(16), 4964–79. https://doi.org/10.1080/00207543.2020.1735666.

Qiao, Weibiao, Kun Huang, Mohammadamin Azimi, & Shuai Han. (2019). A Novel Hybrid Prediction Model for Hourly Gas Consumption in Supply Side Based on Improved Whale Optimization Algorithm and Relevance Vector Machine. *IEEE Access*, 7, 88218–88230. https://doi.org/10.1109/ACCESS.2019.2918156.

Qiu, X., L. Zhang, P. Nagaratnam Suganthan, & G.A.J. Amaratunga. (2017). Oblique Random Forest Ensemble via Least Square Estimation for Time Series Forecasting. *Information Sciences*, 420, 249–62. https://doi.org/10.1016/j.ins.2017.08.060.

Raguseo, Elisabetta. (2018). Big Data Technologies: An Empirical Investigation on Their Adoption, Benefits and Risks for Companies. *International Journal of Information Management*, 38 (February), 187–195. https://doi.org/10.1016/j.ijinfomgt.2017.07.008.

Ramos-Carrasco, Rogelio, Shirley Galvez-Diaz, Victor Nuñez-Ponce, & Jose Alvarez-Merino. (2019). Artificial Neural Networks to Estimate the Forecast of Tourism Demand in Peru. In 2019 IEEE Sciences and Humanities International Research Conference (SHIRCON), 1–4. https://doi.org/10.1109/SHIRCON48091.2019.9024873.

Ray, Papia, Debani Prasad Mishra, & Rajesh Kumar Lenka. (2016). Short Term LoadForecasting by Artificial Neural Network. In 2016 International Conference on NextGenerationIntelligentSystems(ICNGIS),https://doi.org/10.1109/ICNGIS.2016.7854003.

Ren, Yongchang, Tao Xing, Xiaoji Chen, Xu E, & Ying Zhao. (2010). Based on Process Neural Network Learning Algorithm for Prediction of Urban Water Consumption. In 2010 International Conference on Computing, Control and Industrial Engineering, 2, 34–37. https://doi.org/10.1109/CCIE.2010.127.

Riahi, Youssra, Tarik Saikouk, Angappa Gunasekaran, & Ismail Badraoui. (2021). Artificial Intelligence Applications in Supply Chain: A Descriptive Bibliometric Analysis and Future Research Directions. *Expert Systems with Applications*, 173 (July), 114702. https://doi.org/10.1016/j.eswa.2021.114702.

Risi, A. & Pietro, S. (2018). Research on supply chain application based on big data. *International Journal of Smart Business and Technology*, 6(1), 1-14. https://doi.org/10.21742/IJSBT.2018.6.1.01.

Ribeiro, A.M.N.C., P.R.X. Do Carmo, I.R. Rodrigues, D. Sadok, T. Lynn, & P.T. Endo. (2020). Short-Term Firm-Level Energy-Consumption Forecasting for Energy-Intensive Manufacturing: A Comparison of Machine Learning and Deep Learning Models. *Algorithms*, 13(11), 1–19. https://doi.org/10.3390/a13110274.

Romano, Michele, & Zoran Kapelan. (2014). Adaptive Water Demand Forecasting for near Real-Time Management of Smart Water Distribution Systems. *Environmental Modelling & Software*, 60, 265–76. https://doi.org/10.1016/j.envsoft.2014.06.016.

Runge, J., R. Zmeureanu, & M. Le Cam. (2020). Hybrid Short-Term Forecasting of the Electric Demand of Supply Fans Using Machine Learning. *Journal of Building Engineering*, 29. https://doi.org/10.1016/j.jobe.2019.101144.

Sahay, Kishan Bhushan, Suneet Sahu, & Pragya Singh. (2016). Short-Term Load Forecasting of Toronto Canada by Using Different ANN Algorithms. In 2016 IEEE 6th International Conference on Power Systems (ICPS), 1–6. https://doi.org/10.1109/ICPES.2016.7584044.

Saini, Umesh, Rajesh Kumar, Vipin Jain, & M.U Krishnajith. (2020). Univariant Time Series Forecasting of Agriculture Load by Using LSTM and GRU RNNs. In 2020 IEEE Students Conference on Engineering Systems (SCES), 1–6. https://doi.org/10.1109/SCES50439.2020.9236695.

Sajjad, M., Z.A. Khan, A. Ullah, T. Hussain, W. Ullah, M.Y. Lee, & S.W. Baik. (2020). A Novel CNN-GRU-Based Hybrid Approach for Short-Term Residential Load Forecasting. *IEEE Access*, 8, 143759–68. https://doi.org/10.1109/ACCESS.2020.3009537.

Sarhani, M. & A. El Afia. (2015). Intelligent System Based Support Vector Regression for Supply Chain Demand Forecasting. In 2014 2nd World Conference

on Complex Systems, WCCS 2014, 79–83. https://doi.org/10.1109/ICoCS.2014.7060941.

Sasser, W. Earl. (1976). Match Supply and Demand in Service Industries. *Harvard Business Review*, November 1, 1976. https://hbr.org/1976/11/match-supply-and-demand-in-service-industries.

Selvi, M Vetri & Sukumar Mishra. 2020. Investigation of Performance of Electric Load Demand Forecasting with New Architecture Realized in Long Short-Term Memory Deep Learning Network. In 2020 IEEE First International Conference on Smart Technologies for Power, Energy and Control (STPEC), 1–6. https://doi.org/10.1109/STPEC49749.2020.9297785.

Shahrabi, J., E. Hadavandi, & S. Asadi. 2013. Developing a Hybrid Intelligent Model for Forecasting Problems: Case Study of Tourism Demand Time Series. *Knowledge-Based Systems*, 43, 112–22. https://doi.org/10.1016/j.knosys.2013.01.014.

Shao, Y. E., C.-J. Lu, & C.-D. Hou. (2014). Hybrid Soft Computing Schemes for the Prediction of Import Demand of Crude Oil in Taiwan. *Mathematical Problems in Engineering*. https://doi.org/10.1155/2014/257947.

Shareef, Mahmud Akhter, Yogesh K. Dwivedi, & Vinod Kumar. (2016). Exploring Multichannel Design: Strategy and Consumer Behaviour. *The Marketing Review*, 16(3), 235–63. https://doi.org/10.1362/146934716X14636478977674.

Sharifzadeh, M., A. Sikinioti-Lock, & N. Shah. (2019). Machine-Learning Methods for Integrated Renewable Power Generation: A Comparative Study of Artificial Neural Networks, Support Vector Regression, and Gaussian Process Regression. *Renewable and Sustainable Energy Reviews*, 108, 513–38. https://doi.org/10.1016/j.rser.2019.03.040.

Shi, Xiaofeng. (2020). Tourism Culture and Demand Forecasting Based on BP Neural Network Mining Algorithms. *Personal and Ubiquitous Computing*, 24(2), 299–308. https://doi.org/10.1007/s00779-019-01325-x.

Shilaja, C. & T. Arunprasath. (2020). Energy Demand Classification by Probabilistic Neural Network for Medical Diagnosis Applications. *Neural Computing & Applications*, 32(15), 11129-11136. https://doi.org/10.1007/s00521-018-03978-w.

Sidhu, Ravneet Kaur, Ravinder Kumar, & Prashant Singh Rana. (2020). Machine Learning Based Crop Water Demand Forecasting Using Minimum Climatological Data. *Multimedia Tools and Applications*, 79(19–20), 13109–24. https://doi.org/10.1007/s11042-019-08533-w.

Simchi-levi, David, and Yao Zhao. (2005). Safety Stock Positioning in Supply Chains with Stochastic Lead Times. *Manufacturing & Service Operations Management*, 7 (October): 295–318. https://doi.org/10.1287/msom.1050.0087.

Singh, Navneet Kumar, Asheesh Kumar Singh, & Navin Paliwal. (2016). NeuralNetwork Based Short-Term Electricity Demand Forecast for Australian States. In2016 IEEE 1st International Conference on Power Electronics, Intelligent ControlandEnergySystems(ICPEICES),https://doi.org/10.1109/ICPEICES.2016.7853603.

Slimani, Ilham, Ilhame El Farissi, & Said Achchab. (2015). Artificial Neural Networks for Demand Forecasting: Application Using Moroccan Supermarket Data. In 2015 15th International Conference on Intelligent Systems Design and Applications (ISDA), 266–71. https://doi.org/10.1109/ISDA.2015.7489236.

Solyali, D. 2020. "A Comparative Analysis of Machine Learning Approaches for Short-/Long-Term Electricity Load Forecasting in Cyprus." *Sustainability* (*Switzerland*) 12 (9). https://doi.org/10.3390/SU12093612.

Souza, Ricardo Felicio, Peter Wanke, and Henrique Correa. 2020. "Demand Forecasting in the Beauty Industry Using Fuzzy Inference Systems." *Journal of Modelling in Management* 15 (4): 1389–1417. https://doi.org/10.1108/JM2-03-2019-0050.

Spalanzani, Alain, and Karine Evrard Samuel. 2006. "L'absorption de l'incertitude Dans La Chaîne Logistique : Passé, Présent et Futurs." *Logistique & Management* 14 (2): 38–51. https://doi.org/10.1080/12507970.2006.11516864.

Szul, T., K. Necka, and T.G. Mathia. 2020. "Neural Methods Comparison for Prediction of Heating Energy Based on Few Hundreds Enhanced Buildings in Four Season's Climate." *Energies* 13 (20). https://doi.org/10.3390/en13205453.

Tabrizchi, Hamed, Mohammad Masoud Javidi, and Vahid Amirzadeh. 2021. "Estimates of Residential Building Energy Consumption Using a Multi-Verse Optimizer-Based Support Vector Machine with k-Fold Cross-Validation." *Evolving Systems* 12 (3): 755–67. https://doi.org/10.1007/s12530-019-09283-8.

Tanaka, Kenji. 2010. "A Sales Forecasting Model for New-Released and Nonlinear Sales Trend Products." *Expert Systems with Applications* 37 (November): 7387–93. https://doi.org/10.1016/j.eswa.2010.04.032.

Tarafdar, M., C. M. Beath, and J. W. Ross. 2017. "Enterprise Cognitive Computing Applications: Opportunities and Challenges." *IT Professional* 19 (4): 21–27. https://doi.org/10.1109/MITP.2017.3051321.

Teixeira, João Paulo, and Paula Odete Fernandes. 2012. "Tourism Time Series Forecast -Different ANN Architectures with Time Index Input." *Procedia Technology* 5: 445–54. https://doi.org/10.1016/j.protcy.2012.09.049.

Toqué, Florian, Mostepha Khouadjia, Etienne Come, Martin Trepanier, and Latifa Oukhellou. 2017. "Short Amp; Long Term Forecasting of Multimodal Transport Passenger Flows with Machine Learning Methods." In 2017 IEEE 20th International Conference on Intelligent Transportation Systems (ITSC), 560–66. https://doi.org/10.1109/ITSC.2017.8317939.

Tranfield, David, David Denyer, and Palminder Smart. 2003. "Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review." *British Journal of Management* 14 (3): 207–22. https://doi.org/10.1111/1467-8551.00375.

Unsihuay-Vila, C., A. C. Zambroni de Souza, J. W. Marangon-Lima, and P. P. Balestrassi. 2010. "Electricity Demand and Spot Price Forecasting Using Evolutionary Computation Combined with Chaotic Nonlinear Dynamic Model." *International Journal of Electrical Power & Energy Systems* 32 (2): 108–16. https://doi.org/10.1016/j.ijepes.2009.06.018.

UNUTMAZ, Yeşim Esra, Alpaslan DEMİRCİ, Said Mirza Tercan, and Recep Yumurtaci. 2021. "Electrical Energy Demand Forecasting Using Artificial Neural Network." In 2021 3rd International Congress on Human-Computer Interaction, Optimization and Robotic Applications (HORA), 1–6. https://doi.org/10.1109/HORA52670.2021.9461186.

Uyar, Kaan, Ümit ilhan, and Ahmet İlhan. 2016. "Long Term Dry Cargo Freight Rates Forecasting by Using Recurrent Fuzzy Neural Networks." *Procedia Computer Science*, 12th International Conference on Application of Fuzzy Systems and Soft Computing, ICAFS 2016, 29-30 August 2016, Vienna, Austria, 102 (January): 642–47. https://doi.org/10.1016/j.procs.2016.09.455.

Vahdani, Behnam, Farzad Razavi, and S. Meysam Mousavi. 2016. "A High Performing Meta-Heuristic for Training Support Vector Regression in Performance Forecasting of Supply Chain." *Neural Computing & Applications* 27 (8): 2441–51. https://doi.org/10.1007/s00521-015-2015-8.

Van Nguyen, T., L. Zhou, A.Y.L. Chong, B. Li, and X. Pu. 2020. "Predicting Customer Demand for Remanufactured Products: A Data-Mining Approach." *European Journal of Operational Research* 281 (3): 543–58. https://doi.org/10.1016/j.ejor.2019.08.015.

Vanichrujee, Ukrish, Teerayut Horanont, Wasan Pattara-atikom, Thanaruk Theeramunkong, & Takahiro Shinozaki. (2018). Taxi Demand Prediction Using Ensemble Model Based on RNNs and XGBOOST. In *2018 International Conference*

on Embedded Systems and Intelligent Technology International Conference on Information and Communication Technology for Embedded Systems (ICESIT-ICICTES), 1–6. https://doi.org/10.1109/ICESIT-ICICTES.2018.8442063.

Varanasi, Jyothi, and M.M. Tripathi. (2016). "Artificial Neural Network Based Wind Power Forecasting in Belgium." In 2016 IEEE 7th Power India International Conference (PIICON), 1–6. https://doi.org/10.1109/POWERI.2016.8077378.

Venkatesh, Kamini, Vadlamani Ravi, Anita Prinzie, & Dirk Van den Poel. (2014). "Cash Demand Forecasting in ATMs by Clustering and Neural Networks." *European Journal of Operational Research*, 232(2), 383–92. https://doi.org/10.1016/j.ejor.2013.07.027.

Vijai, Praveen, & P Bagavathi Sivakumar. (2018). Performance Comparison of Techniques for Water Demand Forecasting. *Procedia Computer Science*, 8th International Conference on Advances in Computing & Communications (ICACC-2018), 143 (January): 258–66. https://doi.org/10.1016/j.procs.2018.10.394.

Villegas, Marco A., Diego J. Pedregal, and Juan R. Trapero. (2018). A Support Vector Machine for Model Selection in Demand Forecasting Applications. *Computers & Industrial Engineering* 121 (July): 1–7. https://doi.org/10.1016/j.cie.2018.04.042.

Vink, K., E. Ankyu, & Y. Kikuchi. 2020. Long-Term Forecasting Potential of Photo-Voltaic Electricity Generation and Demand Using R. *Applied Sciences (Switzerland)* 10 (13). https://doi.org/10.3390/app10134462.

Wan, Xiang, & Philip T. Evers. (2011). Supply Chain Networks With Multiple Retailers: A Test of the Emerging Theory on Inventories, Stockouts, and Bullwhips. *Journal of Business Logistics* 32 (1): 27–39. https://doi.org/10.1111/j.2158-1592.2011.01003.x.

Wanchoo, Karan. (2019). Retail Demand Forecasting: A Comparison between Deep Neural Network and Gradient Boosting Method for Univariate Time Series. In 2019 *IEEE 5th International Conference for Convergence in Technology (I2CT)*, 1–5. https://doi.org/10.1109/I2CT45611.2019.9033651.

Wang, H., Z. Yang, Q. Yu, T. Hong, & X. Lin. (2018). Online Reliability Time Series Prediction via Convolutional Neural Network and Long Short Term Memory for Service-Oriented Systems. *Knowledge-Based Systems* 159: 132–47. https://doi.org/10.1016/j.knosys.2018.07.006.

Wang, Jian, & Amar Duggasani. (2020). Forecasting Hotel Reservations with Long Short-Term Memory-Based Recurrent Neural Networks. *International Journal of Data Science and Analytics* 9 (1): 77–94. https://doi.org/10.1007/s41060-018-0162-6.

Wang, Y. (2011). Prediction of Weather Impacted Airport Capacity Using Ensemble Learning. In , 2D61–611. https://doi.org/10.1109/DASC.2011.6096002.

Weisbach, Michéle, Utz Spaeth, Milad Ghobadi, & Benedikt Schmuelling. (2020). Power Demand Prediction of Battery Overhead Line Buses Based on a Neural Network Optimization. In 2020 IEEE Green Technologies Conference(GreenTech), 133–35. https://doi.org/10.1109/GreenTech46478.2020.9289715.

Weng, T., W. Liu, and J. Xiao. 2019. "Supply Chain Sales Forecasting Based on LightGBM and LSTM Combination Model." *Industrial Management and Data Systems* 120 (2): 265–79. https://doi.org/10.1108/IMDS-03-2019-0170.

White, Martin. (2012). Digital Workplaces Vision and Reality. *Business Information Review* 29 (December): 205–14. https://doi.org/10.1177/0266382112470412.

Wu, F., S. Hong, W. Zhao, X. Wang, X. Shao, X. Wang, & X. Zheng. 2021. Neural Networks with Improved Extreme Learning Machine for Demand Prediction of Bike-Sharing. *Mobile Networks and Applications*. https://doi.org/10.1007/s11036-021-01737-1.

Xia, Changhao, Jian Wang, & Karen McMenemy. (2010). Short, Medium and Long Term Load Forecasting Model and Virtual Load Forecaster Based on Radial Basis Function Neural Networks. *International Journal of Electrical Power & Energy Systems* 32 (7): 743–50. https://doi.org/10.1016/j.ijepes.2010.01.009.

Xia, Min, Yingchao Zhang, Liguo Weng, & Xiaoling Ye. (2012). Fashion Retailing Forecasting Based on Extreme Learning Machine with Adaptive Metrics of Inputs. *Knowledge-Based Systems* 36: 253–59. https://doi.org/10.1016/j.knosys.2012.07.002.

Xu, Jun, Rouhollah Rahmatizadeh, Ladislau Bölöni, & Damla Turgut. (2018). Real-Time Prediction of Taxi Demand Using Recurrent Neural Networks. *IEEE Transactions on Intelligent Transportation Systems* 19 (8): 2572–81. https://doi.org/10.1109/TITS.2017.2755684.

Xu, Yuebing, Jing Zhang, Zuqiang Long, & Mingyang Lv. (2019). Daily Urban Water Demand Forecasting Based on Chaotic Theory and Continuous Deep Belief Neural Network. *Neural Processing Letters* 50 (2): 1173–89. https://doi.org/10.1007/s11063-018-9914-5.

Xue, Hong, Chi Jiang, Boyu Cai, & Yi Yuan. (2018). Research on Demand Forecasting of Retail Supply Chain Emergency Logistics Based on NRS-GA-SVM. In 2018 Chinese Control And Decision Conference (CCDC), 3647–52. https://doi.org/10.1109/CCDC.2018.8407756.

Yadav, Vineet, and Dipti Srinivasan. (2011). A SOM-Based Hybrid Linear-Neural Model for Short-Term Load Forecasting. *Neurocomputing* 74 (17): 2874–85. https://doi.org/10.1016/j.neucom.2011.03.039.

Yang, Yandong. (2015). Development of the Regional Freight Transportation Demand Prediction Models Based on the Regression Analysis Methods. *Neurocomputing*, 158: 42–47. https://doi.org/10.1016/j.neucom.2015.01.069.

Yang, Yi, Yanhua Chen, Yachen Wang, Caihong Li, & Lian Li. (2016). Modelling a Combined Method Based on ANFIS and Neural Network Improved by DE Algorithm: A Case Study for Short-Term Electricity Demand Forecasting. *Applied Soft Computing* 49 (December): 663–75. https://doi.org/10.1016/j.asoc.2016.07.053.

Yeh, Wei-Chang, Yuan-Ming Yeh, Po-Chun Chang, Yun-Chin Ke, and Vera Chung. 2014. "Forecasting Wind Power in the Mai Liao Wind Farm Based on the Multi-Layer Perceptron Artificial Neural Network Model with Improved Simplified Swarm Optimization." *International Journal of Electrical Power & Energy Systems* 55: 741–48. https://doi.org/10.1016/j.ijepes.2013.10.001.

Yin, L., Z. Sun, F. Gao, & H. Liu. 2020. Deep Forest Regression for Short-Term Load Forecasting of Power Systems. *IEEE Access* 8: 49090–99. https://doi.org/10.1109/ACCESS.2020.2979686.

Yu, Haifeng, Liangyun Song, Weiming Zhu, Xinya Wang, Guozhong Ding, & Shiwei Xia. 2021. Electric Power Material Demand Forecasting Based on LSTM and GM-BP Methods. In 2021 4th International Conference on Energy, Electrical and Power Engineering (CEEPE), 1058–62. https://doi.org/10.1109/CEEPE51765.2021.9475831.

Yu, Xiaodan, Zhiquan Qi, & Yuanmeng Zhao. (2013). Support Vector Regression for Newspaper/Magazine Sales Forecasting." *Procedia Computer Science* 17: 1055–1062. https://doi.org/10.1016/j.procs.2013.05.134.

Zhang, C., J. Li, Y. Zhao, T. Li, Q. Chen, & X. Zhang. (2020). A Hybrid Deep Learning-Based Method for Short-Term Building Energy Load Prediction Combined with an Interpretation Process. *Energy and Buildings*, 225. https://doi.org/10.1016/j.enbuild.2020.110301.

Zhang, G., C. Tian, C. Li, J.J. Zhang, & W. Zuo. (2020). Accurate Forecasting of Building Energy Consumption via a Novel Ensembled Deep Learning Method Considering the Cyclic Feature. *Energy* 201. https://doi.org/10.1016/j.energy.2020.117531.

Zhu, X., S. Yang, J. Lin, Y.-M. Wei, & W. Zhao. (2018). Forecasting China's Electricity Demand up to 2030: A Linear Model Selection System. *Journal of*

Modelling in Management 13 (3): 570-86. https://doi.org/10.1108/JM2-11-2017-0117.